REVIEW ARTICLE

FAIR Geovisualizations: Definitions, Challenges, and the Road Ahead

Auriol Degbelo

Institute for Geoinformatics, University of Münster auriol.degbelo@uni-muenster.de

ARTICLE HISTORY

Compiled November 16, 2021

ABSTRACT

The availability of open data and of tools to create visualizations on top of these open datasets have led to an ever-growing amount of geovisualizations on the Web. There is thus an increasing need for techniques to make geovisualizations FAIR - Findable, Accessible, Interoperable, and Reusable. This article explores what it would mean for a geovisualization to be FAIR, presents relevant approaches to FAIR geovisualizations. The discussion is done using three complementary perspectives: the computer, which stores geovisualizations digitally; the analyst, who uses them for sensemaking; and the developer, who creates them. The framework for FAIR geovisualizations proposed, and the open questions identified are relevant to researchers working on findable, accessible, interoperable, and reusable online visualizations of geographic information.

KEYWORDS

Geovisualization; semantic interoperability; software reuse; FAIR principles

1. Introduction

The momentum gained by the open data movement both in government and research has yielded a large amount of open (geographic) datasets available on the Web. Statistics from Open Data Inception (https://opendatainception.io/) suggest that there are currently more than 2600 open data portals worldwide. If one considers that the average data portal offers thousands of datasets, this means hundreds of thousands of open datasets to be made sense of. Visualization of these datasets has been identified as a key user need in previous work (Degbelo, 2020b; Graves and Hendler, 2013, 2014). Geovisualizations indeed hold the promise of increasing information transparency (Degbelo and Kauppinen, 2018; Marzouki et al., 2017), facilitating data-literate citizenry (Degbelo, Granell, et al., 2016), and catalyzing citizen participation (Fechner and Kray, 2014; Marzouki et al., 2017), ultimately "making the world a better place" (Kraak and Fabrikant, 2017).

Parallel to the developments in the open data landscape, commercial and opensource tools/libraries are increasingly available to create visualizations (e.g. ArcGIS Online, Tableau, D3.js, R). It can thus be expected that (geo)visualizations - next to websites and datasets - will become important contributors to the information overload problem in the digital age. In fact, it can be argued that they are already - witness for example the overwhelming amount of online maps created during the COVID-19 pandemic¹. Finding online maps on a specific topic is becoming increasingly challenging, though solutions are emerging ad-hoc (primarily in the form of online platforms) to provide entry points for them. Example of these platforms include the D3.js Graph Gallery (https://www.d3-graph-gallery.com/), DataUSA (https://datausa.io/), Data-smart city solutions (http://bit.ly/2p23CrS), ArcGIS StoryMaps (https://storymaps.arcgis.com/), Observable (https://observablehq.com/collection/@observablehq/maps), and the R Shiny Gallery (https://shiny.rstudio.com/gallery/), to name a few². The time is ripe to develop techniques making these visualizations first-class in information search processes on the Web.

Safarov, Meijer, and Grimmelikhuijsen (2017) identified users of open government data to include citizens, businesses, researchers, developers, non-governmental organizations, and journalists. All these groups can potentially benefit from techniques that make not only websites/datasets about a topic, but also available geovisualizations FAIR - Findable, Accessible, Interoperable, and Reusable. A question at this point is: why care about making online geovisualizations FAIR at all?

From the theoretical perspective, the uniqueness of web maps as geospatial resources can be highlighted from two perspectives mentioned in (Lai and Degbelo, 2021): map as a *tool* and map as a *representation* of geographic space. First, as a *tool*, web maps are useful to explore the spatial dimensions and interrelationships between phenomena and activities located in space. Thus, they are helpful to create and communicate (visual) stories about geographic phenomena. This storytelling feature is neither present in (raw) datasets nor web services on the spot. Second, as a *form of knowledge representation*, they index information by location in a plane as opposed to using sentences as primary units to organize knowledge. Thus, they enable the retrieval of insight hidden in datasets in a more efficient and effective way. Whether approached from the perspective of a tool that enables story construction, or an artifact that stores geographic knowledge in an effective way, (web) maps are sufficiently distinct from raw datasets and web services to deserve efforts aimed to improve their discoverability.

From the practical perspective, a user survey conducted by Graves and Hendler (2014) has reported evidence of stakeholders' interests in the *reuse* of existing online visualizations. Finding these visualizations in the first place is thus critical to meet these users' needs. Yet, the findability of online (geo)visualizations using existing search engines is currently limited. Exemplar questions where state-of-the-art engines would fail to return appropriate answers include: show *all* online maps about the latest earthquake in Christchurch (e.g. for a geologist); show *all* online maps about the demographics of Berlin, Germany (e.g. for an entrepreneur prospecting about opening a new business), show *all* online maps about the kingdom of Prussia (e.g. for a historian who wants to take advantage of map collections available worldwide about their object of study). This suggests that more work on FAIR geovisualizations is currently needed. This article aims to provide a start to discussions about FAIRness issues pertaining to online geovisualizations. Four questions are addressed:

- What does it mean for a geovisualization to be FAIR?
- What are approaches relevant to FAIR geovisualizations?
- What are unique challenges of FAIR geovisualizations?
- What are open questions for FAIR geovisualizations research?

¹See examples of these maps in (Griffin, 2020). For a non-exhaustive collection, see https://www.lrg.tum.de/lfk/service/online-maps-on-corona-covid-19/ (accessed: March 12, 2021).

 $^{^2\}mathrm{All}$ links were last accessed on July 08, 2021.

'Geovisualization' is a term used with a variety of meanings as indicated by Çöltekin, Janetzko, and Fabrikant (2018). It can denote (i) the process of creating interactive visualizations to aid visual thinking and knowledge building during data exploration; (ii) the artifact (e.g. plots, maps, combinations of these) used to support that process; (iii) the artifact resulting from that process (e.g. a map supplemented with graphical annotations that record users' observations during data exploration); or (iv) the academic discipline concerned with the study of the process and the artifacts. The focus of this article is on findable, accessible, interoperable and reusable *artifacts*.

The two key contributions of the paper are a framework and a research agenda for FAIR online geovisualizations. First, while existing discussions on FAIRness mainly take a computer-centric perspective and focus primarily on technical challenges, a novelty in this work is a holistic discussion of FAIRness. That is, FAIRness is also discussed from the perspective of both the consumer and the producer of the resources to be made FAIR. The discussion of the peculiarities of the three perspectives results into a framework that can be used for (systematic) mapping studies of FAIR geovisualization research. A second novelty of the work is the application of FAIR principles to a new domain, namely that of geovisualizations, to unveil a unique set of challenges and opportunities for GIScience research.

Relevant literature to the questions above is compiled and critically summarized in this article. Since FAIR geovisualizations is an emerging research topic and not a major theme of any established outlet in GIScience, a systematic review would be premature at this stage. Instead, the article draws on exemplar contributions pertinent to the exposition of the ideas. Section 2 summarizes the FAIR principles, and presents a scenario for FAIR geovisualizations. Section 3 briefly reports on existing visions/research agendas and highlights their peculiarities from a metaphor-theoretical point of view. As this section shall clarify, FAIR geovisualizations rely on a slightly different metaphor from prevalent ones in geovisualization research. Section 4 considers the three roles relevant in the context of FAIR geovisualizations (the computer, the analyst, the developer), and elaborates on their uniqueness. Section 5 discusses unique challenges of realizing FAIR geovisualizations. Section 6 brings forth some open research questions on the road towards FAIR geovisualizations, Section 7 discusses limitations, and Section 8 concludes the article.

2. Setting the scene

2.1. What FAIR is

The FAIR principles were originally proposed by M. D. Wilkinson, Dumontier, et al. (2016) in the context of Open Science to emphasize the ability of machines to automatically find and use datasets (or digital assets more broadly), in addition to supporting their reuse by individuals. A wide adoption followed their introduction and with that wide adoption, a variety of interpretations of their meaning emerged. Follow-up articles to the original Wilkinson article tried to clarify the original intent. The following quotes are extracted from some of these.

- "FAIR refers to a set of principles, focused on ensuring that research objects are reusable, and actually will be reused, and so become as valuable as is possible" (Mons et al., 2017).
- "The Principles are aspirational, in that they do not strictly define how to achieve a state of 'FAIRness', but rather they describe a continuum of features, at-

tributes, and behaviors that will move a digital resource closer to that goal" (M. D. Wilkinson, Sansone, et al., 2018).

• "[The Principles] describe characteristics and aspirations for systems and services to support the creation of valuable research outputs that could then be rigorously evaluated and extensively reused, with appropriate credit, to the benefit of both creator and user" (Mons et al., 2017).

From the foregoing, FAIR are a set of aspirational principles introduced initially to improve the reuse of research objects. While the principles are useful to research objects, their scope need not be limited to these. They could indeed benefit other types of resources on the Web, namely online geovisualizations. These online geovisualizations may be the outcome of a research process, or not (e.g. many online geovisualizations are typically created by data journalists, not researchers). Realizing FAIR geovisualizations is thus about ensuring that online geovisualizations are reusable, will be reused, and so become as valuable as possible. In essence, the FAIR principles point at the following key considerations discussed in (Jacobsen et al., 2020):

- Findable: data and metadata must be assigned a globally unique and persistent identifier in order to be found and resolved by computers. In addition, digital resources should be well-described so that they can be accurately discovered.
- Accessible: data and metadata should be retrieved using a standardized communication protocol. An example of standardized access protocol is the Hypertext Transfer Protocol (HTTP).
- Interoperable: data and metadata should be described using a formal language for knowledge representation. (It follows from this definition that interoperability discussions in this article are centered on semantic interoperability aspects, not structural or syntactic aspects).
- Reusable: data and metadata are released with a clear and accessible usage license.

2.2. FAIR Geovisualizations - A scenario

Since there is no prior discussion on FAIR online geovisualizations and no elaborated vision of FAIR geovisualizations in the literature, a thought experiment³ called backcasting is used to frame the current discussion. Following Yeates (2004), backcasting "involves the description of a definite, specific future situation and, then, moving backwards in time, step-by-step, in as many stages as are considered necessary, from the future to the present. The goal of backcasting is to reveal the mechanism through which the specified future could be attained". Backcasting takes a different approach from forecasting. The latter uses current dominant trends to predict what futures are *likely* to happen. Instead, the former examines *desirable* futures, and how they can be attained. While doing so, a backcasting study may question some suppositions inherent in prevailing perceptions, and open some new options (see Dreborg, 1996). Backcasting has been often used for policy planning and analysis (see e.g. Bibri, 2018 for a recent review, and Haslauer, Biberacher, and Blaschke, 2016 for a spatially-explicit backcasting analysis for land-use planning).

³According to Yeates (2004), a thought experiment is a device with which one performs an intentional, structured process of intellectual deliberation in order to speculate about (i) potential antecedents for a designated consequent or (ii) potential consequents for a designated antecedent.

```
<!DOCTYPE html>
<html>
<head>
    <title>Geovisualization Title</title>
        <!-- loading relevant libraries, e.g. Leaflet or D3.js -->
        <script src="..."></script>
</head>
<body>
        Content of the geovisualization.....
        <script type="text/javascript">
        // code (e.g. Leaflet code or D3.js code)
        </script>
```

</body>

</html>

Listing 1: Basic structure of a web-based geovisualization. How to make geovisualizations in this format FAIR?

As an example of *desirable* future⁴, consider the following scenario (Figure 1). A data journalist is preparing a new article on population growth in cities, and is at the exploratory stage. They have no background in Cartography and no programming skills. They go through the following steps: 1) they enter a query in a search engine of their choice; 2) the search engine returns previews and short descriptions of most relevant geovisualizations; 3) the data journalist selects one for closer inspection and realizes (through the grey boxes) that data for the years 2018 and 2019 is missing; 4) they click on the A.M.D. (Add Missing Data) button, and the visualization looks for the missing data on the Web; 5) the data journalist is now interested in visualizing data from another geographic area the same way, to compare patterns of the two regions; 6) a simple drag and drop interaction enables the portability of the current visualizations, of which the generic structure is presented in Listing 1. For a recent review of web-based visualizations, see (Mwalongo et al., 2016).

As the reader may have noticed, this scenario necessitates approaching geovisualizations from three distinct and complementary perspectives. In step 2 of the figure, they are web documents that are being searched (computer perspective). In steps 3 and 4, they are software components that interact with each other (developer perspective). In steps 5 and 6, they are interfaces that are manipulated (analyst perspective). Webbased visualizations support the three perspectives (e.g. through the use of <meta> tags for their description and Javascript to add behaviour). From the theoretical point of view, the following observations (drawing on the distinctions introduced by Larkin and Simon, 1987) can be helpful to relate the three perspectives.

• At a conceptual level, a geovisualization (as an artifact) is a form of representa-

⁴The scenario below is just one of many possible scenarios and is brought forward to advance the discussion on FAIR geovisualizations. Future work may come up with other equally interesting (if not more interesting) scenarios.



Figure 1. FAIR Geovisualizations at work. 1) the user enters a query in a search engine of their choice; 2) the search engine returns previews and short descriptions of most relevant geovisualizations; 3) the user selects one for closer inspection and realizes (through the grey boxes) that data for the years 2018 and 2019 is missing; 4) the user clicks on the A.M.D. (Add Missing Data) button, and the visualization looks for the missing data on the Web; 5) the user is curious to see if they can visualize data from another geographic area the same way; 6) a simple drag and drop interaction enables the portability of the current visualization template to the new dataset.

tion of geographic knowledge. Larkin and Simon (1987) distinguished two ways of representing knowledge in general: sentential (i.e. use sentences as a unit to organize information) and diagrammatic (i.e. index information by location in a plane).

- Online geovisualizations are materialized in several, coexistent forms⁵:
 - Sentential, descriptive: this is the case when they are available as declarative statements of facts (e.g. triples, see Varanka and Usery, 2018). This form entails considerations from the web documents/computer perspective;
 - Sentential, executable: this is the case when they are made available as a software code (e.g. specified using the Vega-Lite grammar, see Köbben and Kraak, 2020). This form entails considerations from the software/developer perspective;
 - Diagrammatic: that is, a visual output (e.g. Chart, Map), which may be interactive or not. This form entails considerations from the interface/analyst perspective.

The three perspectives (computer, analyst, developer) are discussed in detail in Section 4.

3. Related work

Several research agendas and visions have been suggested touching upon the future of geovisualization research. These research agendas and visions overlap to some extent. A means of highlighting their subtle differences is through the use of metaphor theory. As Lakoff and Johnson (2003) indicated, "the essence of metaphor is understanding and experiencing one kind of thing in terms of another".

3.1. Prevalent metaphors in geovisualization research

From existing work, at least three metaphors have inspired the progress of geovisualization research. These metaphors, extracted from (Kent and Klosterman, 2000; MacEachren and Kraak, 2001) are: (i) geovisualization as a *knowledge base* (or a database) storing geographical facts; (ii) geovisualization as an *interface* that can support productive information access and knowledge construction activities; and (iii) geovisualization as a *presentation medium* making a specific point and/or persuading the viewer to adopt a particular position. The knowledge base in (i) can take the form of an *image* encoding geographical facts or a *geometric structure* encoding geographic facts (see Peuquet, 1988).

Existing visions and research agendas have discussed geovisualization as an interface primarily (i.e. metaphor (ii) above). These interfaces can be two-dimensional (as customary with computer screens) or three-dimensional (e.g. a globe or an immersive environment). About 20 years ago, MacEachren and Kraak (2001) listed open questions for geovisualization research, starting with the premise that the map is an interface that can support productive information access and knowledge construction activities while retaining its traditional role as a presentation device. Roth (2013b) defined cartographic interaction as the dialog between a human and a map mediated through a computing device. He discussed the what, why, when, who, where, and how of cartographic interaction, and proposed open questions related to these themes.

⁵The informational and computational equivalence of these three forms are not discussed here.

Çöltekin, Bleisch, et al. (2017) collected and synthesized community input on what experts considered as persistent challenges in geovisualization. Roth, Çöltekin, et al. (2017) discussed methodological aspects of research about user interface aspects (i.e. how to do research about the design and use of interactive maps and visualizations). They also elaborated on research questions in interactive geovisualizations for which more user studies are currently needed.

A few discussions have had a more narrow task focus. Griffin et al. (2017) brought forth research opportunities related to the design of geovisualizations as interfaces that are transferable from one context of use to another. The focus of (G. Andrienko, N. Andrienko, Jankowski, et al., 2007) was on how geovisualizations as interfaces can support spatial decision making, while G. Andrienko, N. Andrienko, Demsar, et al. (2010) discussed how they can support visual analysis. Robinson, Demšar, et al. (2017) looked closely into geovisualization as an interface for the analysis of big spatial data. In (Degbelo and Kray, 2018), the focus is on making geovisualizations more 'intelligent'. The authors brought forth and discussed requirements for intelligent geovisualisations from the perspective of computational user interface design. Finally, Thrash et al. (2019) focused on geovisualisations for individual pedestrians (they use the term geographic information display) and discuss questions of how these can facilitate navigation through a large-scale, real-world environment.

3.2. Another metaphor: geovisualizations as information products

One core premise of this article is that geovisualizations are *information products* in their own right, in a fashion similar to datasets or websites. From the perspective of metaphor theory, the statement 'geovisualizations are information products' is a metaphor, since it involves understanding one domain (i.e. the domain of geovisualizations) in terms of another (i.e. the domain of information products). Metaphors have entailments, through which they highlight and make coherent some aspects of experience.

A key entailment of the statement 'geovisualizations are information products' is that geovisualizations are *gestalts*. A gestalt is "an integrated, coherent structure or form, a whole that is different from the sum of the parts" (Wagemans et al., 2012). That is, **geoviz = geodata + code + interface**, and the whole is more than the sum of its parts. The information product viewpoint implies more than what the knowledge base metaphor (geodata), the interface metaphor (interface) and the presentation medium metaphor (interface) have covered in isolation. Geovisualizations thus deserve their own treatment in information search. Accordingly, dedicated search mechanisms should be developed to make them findable, just as open data portals (e.g. Dataverse, Data.gov, European Data portal) or Google Dataset Search (Noy, Burgess, and Brickley, 2019) have recently emerged to ease the findability of open datasets.

A second entailment of the sentence 'geovisualizations are information products' is that geovisualizations are CONTAINERS (in the metaphor theoretical sense), or in plain language, carriers of informational statements. This entailment is in line with recent proposals to model maps, and more generally visualizations, as a set of statements made by someone at some point in time (see e.g. Degbelo, 2017; Scheider, J. Jones, et al., 2014).

A third entailment of 'geovisualizations are information products' is that geovisualizations are *informational resources* and hence *afford* the FAIR principles from (M. D. Wilkinson, Dumontier, et al., 2016). That is, geovisualizations can (or in principle should) be found, accessed, reused, and afford interaction with other information products on the Web. As MacEachren (1995) aptly indicated: "If we accept the premise that maps ... are a useful way of obtaining spatial information, we have the obligation to *facilitate their use as information sources*" (Page 11, emphasis added). The idea that geovisualizations should be made FAIR is easier to grasp when one looks at static images (arguably one of the simplest forms of geovisualization, and one that is already treated as a specific information product by current search engines).

While geovisualizations afford the FAIR principles, they are yet to be examined in the literature from that perspective. The discussion that follows is an attempt to fill this knowledge gap.

4. FAIR Geovisualizations: three perspectives

The purpose of this section is to provide a conceptual framework for investigating FAIR geovisualizations. As mentioned above, 'geovisualization' can denote an information product, an area of scientific investigation, or a creative process, and the information product viewpoint is of prime interest in this work.

A geovisualization is an artifact, physical or digital, whose visual properties encode geographic data. Roberts (2008) proposed that visualizations of geographic information can take one of seven types: maps/cartograms, networks, charts/graphs, tables, symbols, diagrams, and pictures. In other words, geovisualizations subsume maps. Besides, a further distinction needs to be made between machine-readable and nonmachine readable geovisualization in the context of information search. An example of the former is a web-based visualization created using D3.js, while an example of the latter is a (PNG or JPEG) image. This implies at least 14 types of geovisualizations to be made findable when Roberts' seven types are taken into account. Current search engines have dedicated sections for the search of images (e.g. a map in a GIF format), and as such partly address the findability of static or animated geovisualizations. Solutions enabling the findability of interactive web-based geovisualizations about a given topic will be of increasing value in the coming years.

The application of FAIR principles to geovisualizations requires a clarification of perspectives to remove ambiguities. As Roth (2013b) indicated, cartographic interaction is the dialogue between a map and a human meditated through a computing device. This suggests two key actors affecting a geovisualization during an interaction process: the computer and the human. In addition, the human may be either a producer of the geovisualization or a consumer. In the context of online maps, this distinction matters because the skills required to play these roles are different. A producer requires mastery of a programming language and/or visualization tools (e.g. ArcGIS online, D3.js), while the consumer need not worry about these. There are thus three key roles to consider while investigating FAIR geovisualizations: the computing device (called here 'computer'), the consumer (called henceforth 'analyst' to borrow the terminology of (G. Andrienko, N. Andrienko, Demsar, et al., 2010)), and the producer (called henceforth 'developer'). The developer may be a prosumer (i.e. produce maps without formal cartographic training), or a trained Cartographer as discussed in (Ipatow and Harvey, 2019). Following the scenario (Figure 1), the analyst may be involved in information analysis tasks and/or information search tasks. Realizing FAIR geovisualizations implies coping with demands from a computing perspective and addressing needs of analysts. It also puts some constraints on developers of these geovisualizations. While the exact boundaries demarcating these roles are challenging to formally

pinpoint, the following informal distinction will be used at this point: the computer stores FAIR geovisualizations digitally, the analyst uses them for sensemaking, and the developer creates them.

The concurrent presence of these different roles suggests that there are many dimensions of FAIRness, according to the perspective considered. Yet, the literature so far has only taken a computer-centric view of FAIRness issues. To advance our understanding of these and provide a more holistic view, the three perspectives are considered at this point. Interpretations of FAIR according to the role considered, relevant approaches to address the issues arising for each role and scientific communities of interest are considered below. The discussion is done from a knowledge representation perspective (computer); from the perspective of user interface design (analyst); and from the perspective of technical considerations about the publishing of online geovisualizations (developer). Table 1 summarizes the key takeaways. Since each of the keywords or approaches mentioned in the table would need extensive literature reviews to cover them appropriately, the discussion only points at examples of relevant work and recent reviews, where available.

			Computer's Demands Analyst's Needs		Developer's Constraints		
Ī	\mathbf{F}	Definition	persistent + described	highlighted	aliased		
		Example approaches	digital object identifiers, container-	query auto completion; relevance	query expansion		
			description) automatic- and semi-	attention-guiding visualization			
			automatic description.	attention galaring vibratization			
		Relevant to	Digital Libraries; Semantic Web	Geographic Information Retrieval	GIScience/Geospatial Semantic Web		
ĺ				·			
Α		Definition	compliant (with a standardized	tailored (to abilities and expertise)	platform- and device-agnostic		
			communication protocol)				
Ì		Example approaches	https, ftps	accessible design	map plasticity		
Ì		Relevant to	Web Science	Human-Computer Interaction	Human-Computer Interaction		
ľ							
	Ι	Definition	formalized	compliant (with widespread design conventions)	compliant (with web mapping standards)		
		Example approaches	map as statements/knowledge base	symbol stores; map-, text-, and colour brewers	wms, wfs, wcs, sld, html5, css3, svg, rdf/owl, mapML, webGL, x3d/x3dom		
		Relevant to	GIScience/Cartography & GI- Science/Geospatial Semantic Web	GIScience/Cartography	GIScience/Cartography & GIScience/SDI; Se- mantic Web		
Ī	R	Definition	licensed (under an open and	transformable	modular + open-source + backward-		
			machine-readable licence)		compatible		
		Example approaches	JSON hashes, RDF	template-based design	declarative frameworks, shelf construction		
					tools, geodocuments, blockchain		
		Relevant to	Semantic Web; Open Data	GIScience/Geovisualization & Informa-	GIScience/Geovisualization & Information Vi-		
				tion Visualization	sualization		

Table 1. A framework for FAIR geovisualizations. The framework features three perspectives: demands from a computing perspective, analyst's needs, and constraints for developers of these geovisualizations. The table also shows examples of approaches to realize these demands/needs/constraints. Finally, the table lists exemplar disciplines and communities that investigate and address the demands/needs/constraints.



Figure 2. Interdependences between the dimensions of FAIR geovisualizations. The analyst's demands include many of the computer's demands, and the demands of both computer and analyst result in constraints for the developers. Legend: licenced = licenced with a machine-readable open licence; compliant (c) = compliant with standardized communication protocols; compliant (d) = compliant with widespread design conventions; compliant (m) = compliant with web mapping standards. Some aspects are highlighted in black to indicate that there are optional to realize interoperability and reuse from the user's viewpoint.

4.1. FAIR Geovisualizations - The computer's demands

Findable, from the computer's perspective, means 'persistent' so that resources can be located over and over again. Web evolution (e.g. API interfaces updates, webpages redesign, server relocation) is the main cause for non-persistence of web urls. Studies on the dynamics of the Web have revealed that death rates for web pages are 20%after six months (Koehler, 1999). Death rates for web sites are 12% after a six-month period (Koehler, 1999), and death rates of Linked Data documents amount to about 5% for a six-month period (Käfer et al., 2013). Persistence is currently addressed through the use of digital object identifiers (DOIs) to uniquely identify resources. and containerization to encapsulate computational components. Docker is currently the dominant platform for containerization and offers an open-source specification for defining and running multi-container applications (i.e. Docker Compose). Applications based on Docker Compose can be easily deployed on the Web, but the adoption of the technology has been relatively slow, perhaps due to the learning curve required to master it. In addition, since metadata is key to any effective search, findable from the computer's viewpoint also connotes 'described'. Metadata can be generated manually (e.g. place name tags or full-text description), automatically, or semi-automatically. The question of how to produce these metadata has been investigated for scanned maps in the context of digital libraries (e.g. Kiser and Smeltekop, 2018), and in the context of designing geoportals (e.g. Brovelli et al., 2012). FAIR geovisualizations can learn from, and build upon lessons learned by these and other works.

Accessible, from the computer's viewpoint, means 'compliant with a standardized communication protocol'. As discussed in (Jacobsen et al., 2020), the purpose of this principle is to provide a predictable way for agents to access resources. Ideally, the protocol should be open, free, and universally implementable. It should allow for authentication and authorization when resources have access restrictions due to ethical, legal or contractual constraints. HTTPs and FTPs are examples of protocols underlying the Web that are compliant with these requirements.

Interoperable, from the computer's perspective, means encoded in a formal language for knowledge representation ('formalized' for short). A relevant approach here is the idea of map as statements made by somebody somewhere at some time (Scheider, J. Jones, et al., 2014), also called maps as knowledge base in (Varanka and Usery, 2018). An early work relying on metadata encoded in the Resource Description Framework (RDF) to improve thematic map search was presented in (Aditya and Kraak, 2007). More recently, Scheider, Degbelo, et al. (2014) illustrated how RDF can be used to describe both the properties of a historical map (e.g. map scale) and its content (i.e. cities, buildings and hills shown on the map). Degbelo (2017) showed how the Schema.org vocabulary, which is endorsed by several search engines (e.g. Google, Microsoft, Pinterest, Yandex), can be used to annotate web maps for subsequent search. Gao, Janowicz, and D. Zhang (2016) proposed a map legend ontology, and illustrated how the ontology can help answer questions such as "which map legends contain highway information"? or "which maps contain both ski areas and camping areas?" Contrary to the works mentioned above, which aimed at describing geovisualizations as finished information products in order to improve their search, Gould and Mackaness (2016), and Huang and Harrie (2019) focused on describing the process of producing a geovisualization. The former partly formalized steps of the cartographic generalization process, while the latter formalized the rendering of geometries using ontologies and rules. Finally, MARC (Avram, 2003; Library of Congress, 2020) is a format that has been used by projects digitizing paper maps (e.g. Kiser and Smeltekop, 2018) in the context of digital libraries. It may be extended, or some of its metadata fields (e.g. projection, type of cartographic material) reused for the description of online web-based geovisualizations.

Reusable, from the computer's viewpoint, means that there is a machine-readable open license available that can be accessed ('licensed' for short). Initiatives to advance work in this area come from the open data community, and include the Open Data Commons' License Service⁶ and the RDF Creative Commons Licenses⁷. The Open Data Commons License Service provides licenses in JSON and an API friendly form, while the RDF Creative Commons Licenses helps describe copyright licenses in RDF.

4.2. FAIR Geovisualizations - The analyst's needs

FAIRness needs from the user's perspective are different from the demands from the computing perspective. For instance, from the analyst's point of view, Findable means 'highlighted'. The issue of finding a needle in a haystack is thorny, precisely because the needle resembles all have around. Highlighting is implemented in various contexts in different ways. To cope with the issue that all queries resemble themselves (syntactically), search engines use an auto-completion feature to highlight queries that are likely to better represent users' intent, and help them avoid spelling mistakes and save time typing. In addition, human-computer interaction and map-making rely on user interface elements arrangement - visual hierarchy and layout (see Tait, 2018) - to create a ranked order of visual elements so that the most important elements have the greatest visual prominence. For example, a menu item immediately visible on a web page is easier to find that one that needs to be accessed only after browsing. Thevenin and Coutaz (1999) call the former menu item 'first-class' and the latter 'second-class'. Finally, relevance ranking is also a means of highlighting items of the information space. It puts some items on top and leaves others at the bottom, signaling to the user where their attention should be directed in priority. Hu et al. (2015a,b) used a regression model incorporating nine aspects of matching between input queries and map metadata (e.g. exact or similarity matching on title, description, and place names) to formalize relevance for map search. Their model was applied to the semantic search of maps in ArcGIS online. Swienty et al. (2008) proposed attention-guiding geovisualization to help users quickly locate and easily decode relevant information through the use of suitable graphical variables. They reported promising results about a computational model to guide users' visual attention to most relevant features during the interaction with a geovisualization.

Accessible, from the analyst's viewpoint, means 'tailored to the user's abilities and expertise'. It is strongly related to Shneiderman (2000)'s idea of 'universal usability': break technology barriers, bridge knowledge gaps, and accommodate user diversity. Example approaches to broaden the accessibility of cartographic products include: the use of multi-layered interfaces to bridge the knowledge gap between novice and experts (see Roth, 2013b); and the use of multi-modal interaction to engage a multiplicity of user capabilities in the computer-human communication process (see Obrenovic, Abascal, and Starcevic, 2007). Challenges of accessible maps were recently discussed in (Froehlich et al., 2019), and recommendations to make web maps more accessible were suggested in (Hennig, Zobl, and Wasserburger, 2017). From the theoretical point of view, Griffin et al. (2017) presented a model for operationalizing map use contexts.

⁶https://opendatacommons.org/ (last accessed: March 12, 2021).

⁷https://github.com/creativecommons/cc.licenserdf (last accessed: March 12, 2021).

The model has four key elements: the *user* (e.g. individual differences, capabilities), the *environment* (e.g. the setting of the map use), the *activity* (e.g. purpose for using the map, actions undertaken while using the map), and the *map* (e.g. representation design, nature of data, display device). Accessible design, as discussed here, primarily addresses the diversity of *user abilities*.

Interoperable, from the analyst's viewpoint, means that the geovisualization 'uses widespread design conventions'. Following Roth, Brewer, and Stryker (2011), a user interacts primarily with four aspects of map features: content, geometries (i.e. points, lines, polygons), symbols, and labels. Widespread adoption of conventions is particularly relevant for symbols (map symbology) and labels (map typography). As regards symbology for topographic maps, organizations such as the OrdnanceSurvey⁸ the National Park Service⁹, the USGS¹⁰, and communities such as OpenStreetMap¹¹ have proposed catalogs of symbols that can be used for application domains such as tourist and leisure information, as well as general-purpose mapping. Reusing symbols from these catalogs can increase interoperability from the user's point of view, since this increases their chances of interacting with 'familiar' symbols. Besides, previous work (e.g. Robinson, Pezanowski, et al., 2013) has brought forth the concept of a symbol store to promote the sharing of map symbols in a specialized domain (e.g. emergency management). Next to topographic maps, symbology for thematic maps needs to specify design conventions for the use of Bertin's visual variables during the encoding of geographic data. Approaches to democratize available knowledge regarding such design conventions include (i) recommendations from the literature (e.g. scientific papers, textbooks), and (ii) the use of tools encapsulating (cartography) expertise. As to (i), recommendations for quantitative geodata transformations and effective visualization are available for example in (Kraak, Ricker, and Engelhardt, 2018; Kraak, Roth, et al., 2020). Guidance for labeling and text placement are available online¹² and in the literature (e.g. Guidero, 2017). As to (ii), Colorbrewer (Harrower and Brewer, 2003), the Map Symbol Brewer (Schnabel, 2005), and the TypeBrewer¹³ are examples of tools suggested to help (web) map makers select/create pertinent color schemes, map symbols, and typographic palettes respectively. A brewer is a tool that uses mapping principles to offer users a selection of choices for a cartographic representation challenge (Brewer, 2003). ArcGIS and QGIS also provide options to select color/map/typographic palettes during the creation of digital maps. Again, the availability of these tools does not guarantee interoperability per se. Interoperability is truly realized if the symbols provided by all these catalogs are more or less similar, to minimize the element of surprise on the user's side. Collecting empirical evidence on (i) the extent to which the available symbols/palettes are used, and (ii) the extent to which they differ across providers is an open question for research (see Section 6.3).

Reusable, from the analyst's viewpoint, means 'transformable' (or controllable). With respect to the theoretical model of map use contexts from Griffin et al. (2017), reuse from the user's point of view - as discussed here - primarily addresses the diversity of *user activities*. A relevant idea from Griffin et al. (2017) is the notion of design

⁸See https://www.ordnancesurvey.co.uk/mapzone/assets/doc/Explorer-25k-Legend-en.pdf (last accessed: March 12, 2021).

⁹See https://github.com/nationalparkservice/symbol-library/ (last accessed: March 12, 2021).

¹⁰See https://pubs.usgs.gov/gip/TopographicMapSymbols/topomapsymbols.pdf (last accessed: March 12, 2021).

¹¹See https://wiki.openstreetmap.org/wiki/SymbolsTab (last accessed: March 12, 2021).

¹²E.g. https://www.arcgis.com/apps/Cascade/index.html?appid=e739c503a1f04d38839834a0fe4ca6d4 (last accessed: Marc 12, 2021).

 $^{^{13}}$ At the moment of this writing, TypeBrewer is undergoing an update, and is temporarily offline.

transferability, that is, the exploitation of one or more aspects of an existing design for a new application or map use situation. A way of realizing design transferability from a user's perspective is through the use of *templates* (a.k.a. meaningful defaults). These templates are usually available as a feature of a toolkit, and several toolkits have been proposed in the past, for example, Tableau Public¹⁴, IBM ManyEyes (Viegas et al., 2007) launched in 2007 and closed in 2015, the GaV Toolkit (Van Ho et al., 2012), the Geoviz toolkit (Hardisty and Robinson, 2011), the Geospatial Visual Analytics Toolkit for movement data (N. Andrienko and G. Andrienko, 2008), and AdaptiveMaps (Degbelo, Sarfraz, and Kray, 2020) for the visualization of open geographic data.

4.3. FAIR Geovisualizations - The developer's constraints

The demands from the computing and user perspective bring about constraints for a developer of FAIR geovisualizations. For example, a constraint imposed by Findable visualizations on the developer is that they should be 'aliased' (i.e. items are being referred to in more than one way). Furnas et al. (1987) found that the probability of two people spontaneously picking the same term to refer to an entity is less than 20%, and called this issue the 'vocabulary problem'. Recent studies have found equally low rates. Degbelo, Trilles, et al. (2016) found that the probability of two providers naming categories for open data search in a similar way is less than 30% for European open data catalogs; Feitelson et al. (2020) reported a probability of less than 7% regarding developers picking similar variable and function names in their study. As a cure to the vocabulary problem, Furnas et al. (1987) suggested that developers use an unlimited number of aliases. Query expansion is the main technique used to mitigate the vocabulary problem. It can be done using (i) terminological associations learned from users' past queries, or (ii) a knowledge base. The former approach has been called 'adaptive indices' (see e.g. Furnas, 1985), and the latter typically uses a geographic ontology (e.g. Cardoso and Silva, 2007; Mai et al., 2020), a domain ontology (e.g. Jiang et al., 2018), or both (e.g. Fu, C. B. Jones, and Abdelmoty, 2005; Hu et al., 2015a).

A constraint imposed by Accessible geovisualizations on developers is that they should be 'platform- and device-agnostic'. Users can now interact with geovisualizations on different devices (e.g. desktop computers, smartphones, and large-scale displays), using different interaction modalities (touch, eye-movements, sound, freehand gestures, speech). The challenge for the developer is to provide geovisualizations that preserve functionalities and user experience across all these. The idea of interface plasticity (Thevenin and Coutaz, 1999) and its adaptation to the context of geographic information map plasticity (Kray and Degbelo, 2019) were proposed to mitigate this issue. Plasticity, as discussed here, primarily addresses the diversity of user environments (Griffin et al., 2017). Designing for plasticity, also called 'model-based design? (Meixner, Calvary, and Coutaz, 2014), advocates a separation of concerns, so that an interactive system can withstand variations due to the physical characteristics of devices and properties of the software environments. An *abstract model* describes the content of a geovisualization on an abstract level, independent of how it is ultimately rendered to the user. This abstract model is then translated into one or many concrete *geovisualizations*, which respectively fit the constraints imposed by the user's platform (e.g. Android vs iOS) and interaction device (e.g. a tablet or mobile phone). In the

¹⁴Tableau Public is sometimes classified as a shelf-configuration user interface (UI) rather than a template editor UI (see e.g. Grammel et al., 2013; Satyanarayan, Lee, et al., 2020). Here, Tableau Public is mentioned as an example of template-based design software in reference to the "Show Me" feature that suggests meaningful charts during the visualization creation process.

context of user interface design, the abstract model has been typically encoded in a machine-readable format, for instance, XML (see Lacoche et al., 2019), JSON (see Badam and Elmqvist, 2019) or the UsiXML (USer Interface eXtensible Markup Language)¹⁵. Badam and Elmqvist (2019) presented BusinessVis, a plastic visualization of the Yelp academic dataset. BusinessVis is web-based and has three views connected through brush-and-link interaction: a map view of businesses, a category treemap, and a rating view showing a list of companies and their user ratings; it supports automated rendering of a JSON-based abstract geovisualization model on both a large display and tablets/smartphones.

A constraint imposed by Interoperable geovisualizations on developers is that they should be 'compliant with (open) web mapping standards'. (Standards here includes both de facto and de jure standards.) In the context of web-based geovisualizations, two types of standards are particularly relevant, namely those from the OGC (Open Geospatial Consortium), and those from the W3C (World Wide Web Consortium). Examples of OGC standards relevant to geovisualizations include the Web Map Service ([WMS] for retrieval of map images), the Web Feature Service ([WFS] for retrieval of geographical features), the Web Coverage Service ([WCS] for retrieval of raster data) and the Styled Layer Descriptor ([SLD] for symbolization and coloring of feature and coverage data). Examples of web standards include HTML5 (for web content structuring and presentation), CSS3 (for styling), and the Scalable Vector Graphics (for resolution-independent graphics). Work is ongoing to bridge the two worlds. At the institutional level, the OGC and the W3C have partnered between 2015 and 2017 to develop a joint note on best practices for the publishing of spatial data on the Web¹⁶. At the conceptual level, Janowicz et al. (2010) proposed a semantic enablement layer for spatial data infrastructures to support the discovery of geospatial content. At the technical level, different adapters were proposed to bridge between the two worlds. These adapters typically support the translation of WFS requests into SPARQL queries (e.g. Vilches-Blázquez and Saavedra, 2019), the reverse translation of SPARQL queries into WFS requests (e.g. Zhao, C. Zhang, and Li, 2016), or both (e.g. J. Jones et al., 2014). Next to the bridging between different open standards, interoperability between different geovisualization environments is also an important matter of concern. To this end, Seo and Yoo (2020) proposed an information model to enable interoperability between geovisualizations in geographic, virtual reality (VR), and augmented reality environments (AR). The information model extends HTML with custom element tags to support the rendering of 2D/3D and VR/AR geo-applications in the browser. Another line of work is developing the Map Markup Language (MapML), a text format for encoding map information for the World Wide Web¹⁷. The key objective of MapML is to "identify the Web map processing that is currently performed by JavaScript libraries which should instead be defined as elements and attributes supported by CSS^{"18}.

A constraint imposed by **R**eusable geovisualizations on developers is that they should be 'modular and open-source'. Modularity is well covered by several web mapping libraries (e.g. R, Leaflet, D3.js). A key idea to facilitate reuse from the developer's perspective is that of *declarative frameworks*. Declarative programming languages specify what the results of a computation should be instead of how the results should be computed. They target programmers for what Bostock and Davies (2013) called 'code

¹⁵See http://www.usixml.org/ (last accessed: March 12, 2021).

¹⁶See the announcement about the collaboration at https://www.w3.org/2015/01/spatial.html.en, and the note at https://www.w3.org/TR/sdw-bp/ (both last accessed: March 12, 2021).

¹⁷See https://maps4html.org/ (last accessed: March 12, 2021).

¹⁸https://github.com/Maps4HTML/MapML-Proposal (last accessed: March 12, 2021).

as Cartography'. Besides, some visualization authoring tools provide developers less conversant in programming with decent expressiveness during the visualization creation process. The Information Visualization literature (Grammel et al., 2013; Satyanarayan, Lee, et al., 2020) distinguishes between shelf construction tools (which enable authors to map data fields to visual variables) and visual builders (where authors make use of fine-grained user interface elements - marks, glyphs, coordinate systems, and layouts - for more expressiveness during the visualization creation process). Examples of declarative languages for (geo)visualization design include Protovis (Bostock and Heer, 2009), its successors D3 (Bostock, Ogievetsky, and Heer, 2011) and Vega-lite (Satyanarayan, Moritz, et al., 2017). Examples of shelf construction tools include Tableau Public (which evolved from Polaris: Stolte, Tang, and Hanrahan, 2002) and Voyager (Wongsuphasawat et al., 2016). Examples of visual builders include research prototypes such as Lyra (Satyanarayan and Heer, 2014) and iVizDesigner (Ren, Hollerer, and Yuan, 2014). Furthermore, another important aspect of reuse from the developer's viewpoint is licensing, and at least two useful ideas can be found in existing work: the use of geo-documents, and the use of blockchains. Döllner (2005) proposed to create geovisualizations as geo-documents that embed digital rights about their components. A geo-document viewer is then in charge of rendering those components of a geovisualization that a user should see (e.g. interaction or animation elements), and hide those they are not allowed to see. Y. Zhang et al. (2020) proposed a decentralized approach to digital rights management for spatial datasets, and implemented the approach using the open-source blockchain Ethereum. Key features of their approach include: support for digital rights registration, querying, as well as transactions where a software agent asks and explicitly obtains permission before using a spatial dataset; support for web technologies; and support for both raster and vector datasets. Finally, since web mapping libraries used to build online geovisualizations often undergo changes, backward-compatibility of components is an important aspect of reuse from the developers' viewpoint.

4.4. Summary: what does it mean for a geovisualization to be FAIR?

This section has tried to specify what FAIR means for geovisualizations and reported briefly on existing approaches relevant to making geovisualizations FAIR. Two lessons can be learned from the previous sections and Table 1. First, 'FAIR' in the context of geovisualization can mean different things depending on the perspective adopted. The twelve senses proposed in Table 1 can thus be used by researchers to disambiguate and clarify their standpoint. Second, FAIR geovisualizations provide great potential for interdisciplinary research as their contributions can be relevant to several disciplines and communities.

Though the aspects of FAIRness have been discussed separately to highlight their unique demands, they are interdependent. Figure 2 captures the interdependences. The figure illustrates two key points. First, the concepts of FAIRness involved are multidimensional and the number of dimensions is perspective-dependent as said just above. Second, many of the analyst's demands include the computer's demands to capture the fact that FAIRness from the user's point of view necessitates a solid technological base. The computer's demands and the analyst's needs bring about constraints for the developer.

As to the usefulness of the framework from Table 1, it can be used to systematically map research on FAIR geovisualizations as it evolves. To illustrate the idea, articles on FAIR geovisualization were downloaded using Taylor & Francis Online¹⁹ and the Wiley Online Library²⁰. These two online catalogs were used because they index the majority of GIScience journals (see the list of journals in the supplementary material). 202 articles were returned from a search done in January 2021 using the keywords: 'findable map OR findable geovisualization'; 'accessible map or accessible geovisualization'; 'interoperable map or interoperable geovisualization'; and 'reusable map or reusable geovisualization'. After removing duplicates, review/editorial/encyclopedia articles, 40 articles were found relevant to the issues of knowledge representation, user interface design, or technical considerations of publishing geovisualizations discussed above. The distribution of the topics of these articles is shown in Figure 3. As the figure suggests, the coverage of current research is uneven at the moment. Research on interoperability is well-represented in the sample, and reflects the fact that GI-Science work has focused on ontology/knowledge base development, development of empirically-derived guidelines for map design, and development of visualizations based on existing standards (e.g. WMS and WFS). Perhaps unsurprisingly, research on accessibility from the computer perspective is not represented in the sample, as this does not fall within the core of GIScience work. However, research proposing solutions on machine-readable licenses, advancing aliasing to improve semantic search, or highlighting platform/device-independence of geovisualizations has been relatively scarce so far, and could perhaps receive more attention. Including articles from conference proceedings may alter the picture painted here. Thus, these comments on the current coverage of FAIR geovisualization research are tentative. They serve an illustrative purpose for the value of the framework from Table 1 only. The data for the figure is available as a supplementary material (Section 8).

5. Unique challenges

Having specified how FAIR principles can be applied to geovisualizations, I now turn to the challenges of realizing FAIR geovisualizations. As we shall see, FAIRness applied to web-based geovisualizations presents unique issues for GIScience research. As discussed in Section 4, a geovisualization can come in several formats, for example, raster or web documents. Geovisualizations in raster format typically appear in the context of scanned (historical) maps and some challenges they pose have been discussed extensively in previous work (e.g. Chiang, Duan, et al., 2020; Chiang, Leyk, and Knoblock, 2014). The focus of this section is on web-based visualizations. Web visualizations sometimes contain SVG elements, and Battle et al. (2018) have shown that their types (e.g. bar chart, pie chart, map) can be identified with reasonable accuracy (86%). The challenges of making online web-based geovisualizations FAIR are summarized in Table 2.

5.1. The computer's demands

As discussed in the previous section, findability necessitates persistence. Given that web-based geovisualizations are composite products, their persistence is dependent on the persistence of their components, notably datasets and libraries. It follows that a persistence strategy for geovisualizations needs an explicit account of these dependen-

¹⁹https://www.tandfonline.com/

²⁰https://onlinelibrary.wiley.com/



Figure 3. Coverage of research on FAIR geovisualization as of January 2021.

	Computer	Analyst	Developer			
F	persistence is not standalone, but dependent on multiple components (data and li- braries); provenance descrip- tion involves both aspects of data and aspects of pro- cessing (e.g. projection, map- ping libraries); geovisualiza- tion descriptions need to com- ply with requirements that come with legislation; geovi- sualizations are composed of many data views	relevance should integrate unique human factors (e.g. perceptual effectiveness, readability); geovisual- izations present multiple perspectives on the data; geovisualizations have multi- ple usages on the Web	aliasing, applied to geovisual- ization content, has no exact limits			
A	none	users' abilities and expertise are unknown a priori; ability and expertise are multifaced concepts	device properties (e.g. size, input modalities) are limiting factors to consistent cross- device functionality develop- ment			
-						
1	content modelling as state- ments has no exact limits, be- cause a geovisualization can communicate many messages at a time; geovisualizations inherently use spatial scale as an organizing dimension for content	empirically-derived guide- lines are lacking; not all guidelines are amenable to brewers	there is a large number of standards to master; proto- cols to establish mutual un- derstanding during content exchange are still lacking			
D	moniquelization liconges and	the design grade of georging	defining anticipating range			
ĸ	inherited, and licensing poli- cies of datasets/libraries may differ	alizations is complex (e.g. it involves data attributes, data dimensions, and visual vari- ables)	empirical evidence about reuse needs of geovisualiza- tion developers is lacking			

Table 2. Unique challenges of FAIR geovisualizations from the perspective of knowledge representation (computer), user interface design (analyst) and technical considerations about the publishing of online geovisualizations (developer).

cies. In addition, findability of geovisualizations necessitates their description. Description touches primarily on provenance. Aspects of provenance for web pages may include their author, creation date, currency, contact information, and topic. Likewise, aspects of provenance for datasets could include author, creation date, spatial and temporal coverage. A unique feature of geovisualization is that describing them includes data & web page provenance aspects, but needs to go beyond them to include *aspects of processing* (e.g. map projection used, and web mapping libraries involved in the development of the geovisualization). Furthermore, geovisualizations, qua online products, need descriptions compliant with requirements that come with legislation, e.g. the Americans with Disabilities Act²¹, or the EU's Web Accessibility Directive (which in turn references the Web Content Accessibility Guidelines)²². At last, geovisualizations package several views on the data. These views take several forms discussed in (Degbelo, 2020a): map, data table, bar chart, pie chart, histogram, Q-Q plots, stacked bars, adjacency diagrams, tree-maps. Rich, fine-granular descriptions need to account for all of these.

Interoperability requires the use of machine-readable formats. Following Scheider, J. Jones, et al. (2014) who distinguished between context (e.g. map scale, spatial extent) and content-based aspects of a geovisualization (i.e. statements that users can extract from the visualization by looking at it), machine-readability can touch on both aspects. Simple examples of describing context and content aspects were presented in (Scheider, J. Jones, et al., 2014). Yet, the key challenge, when it comes to representing content formally lies in the fact that a geovisualization may provide multiple perspectives (e.g. a bar chart view or a map view) on a given dataset. To this multiplicity of perspectives should be added the different types of meanings conveyed by geovisualizations. As MacEachren (1995) pointed out, the meaning of a map can be approached from two angles: denotative and connotative. Denotative meaning is either precisely specified in a map legend or assumed to be part of the normal reader's general map schema; connotative meaning is implicitly conveyed by the map (e.g. connotation of veracity and connotation of integrity). It remains unclear at the moment how these different types of meanings can be represented and linked formally in a geographic knowledge base, for the purpose of (semantic) search. At last, as discussed in (Degbelo, 2020a). statements about the content of geovisualizations are granularity-dependent (e.g. some facts pertain only to a given zoom level on a map). This dependence of statements on spatial granularity and the inherent use of spatial scale as an organizing dimension is a distinguishing feature of geovisualizations from other online web documents. Capturing this granularity-dependence in knowledge graphs also presents a unique challenge for semantic interoperability research.

As to reuse, geovisualization licenses are inherited from those of their components. Since licensing policies of datasets/libraries may differ, a general statement about a geovisualization's reuse policy becomes challenging. Tools to automatically reason about copyright policies of the sub-components and inform about the copyright of the final geovisualization will become increasingly valuable.

²¹https://www.ada.gov/pcatoolkit/chap5toolkit.htm (last accessed: July 23, 2021).
²²https://eur-lex.europa.eu/eli/dir/2016/2102/oj (last accessed: July 23, 2021).

5.2. The analyst's needs

The primitive spatial query²³ in the context of geographic information retrieval has the form 'what, relationship, where' (Cardoso and Silva, 2007: Fu, C. B. Jones, and Abdelmoty, 2005) or '<theme><relationship><location>' (Purves, Clough, C. B. Jones, Hall, et al., 2018). In the context of geodata search, the primitive query takes the form 'dataset about, what, where, when' or 'dataset about <space><time><theme> (Degbelo and Sherpa, 2020). The key difference between geographic information retrieval and geodata search is that the former takes unstructured text as input, while the latter takes a structured description of data items (e.g. in a JSON format) as input. Since web-based geovisualizations are based on HTML that is structured. finding them shares some similarities with dataset search. The primitive query here takes the form 'geovisualization about, what, where, when', or 'geovisualization about <space><time><theme>'. Nonetheless, modeling the notion of relevance in this context poses fundamentally new challenges. First and foremost, relevance modeling should integrate perceptual effectiveness, i.e. the ability of a person to retrieve data presented in a chart by decoding a visual representation. Graphical perception studies were conducted to assess, among other things, the effectiveness of bar charts and pie charts (Cleveland and McGill, 1984), treemaps and wrapped bars (Mylavarapu et al., 2019), line charts and scatterplots (Saket, Endert, and Demiralp, 2019), small multiples for time series (Javed, McDonnel, and Elmqvist, 2010), the differences between interactive tables and interactive geovisualizations (Degbelo, Wissing, and Kauppinen, 2018), and alternative layouts for online map design (Cöltekin, Heil, et al., 2009). These have revealed a strong task-dependency effectiveness for geovisualizations, and general guidelines for visualizations' perceptive effectiveness have not yet emerged. Next to perceptual effectiveness, visual map complexity and map readability are important to the interaction with geovisualizations. The literature has come up with preliminary measures for visual map complexity (Schnur, Bektas, and Cöltekin, 2018) and map readability (Harrie, Stigmar, and Djordjevic, 2015). How these could factor in relevance remains an open question. At last, a relevance ranking scheme for online geovisualizations needs to deal with the fact that they have multiple usages. For instance, online geovisualizations can also be used as spatial dialogue platforms (e.g. 'Dialog Map', see Fechner and Kray, 2014), platforms for spatial collaboration and data collection (e.g. Ethermap, see Fechner, Wilhelm, and Kray, 2015), tools to persuade (see Muehlenhaus, 2013), or media to display and refine search results. Retrieving online geovisualizations (e.g. thematic maps) that communicate facts about topics is the main purpose of steps 1-2 of the scenario (Figure 1). Finding means to model and distinguish purposes of online geovisualizations is also a unique research challenge.

Accessibility implies adaptation to the user's abilities and expertise. The key challenge here is that "for maps presented on the Web, the audience is very much unknown" (Köbben and Kraak, 2020). Froehlich et al. (2019) listed two high-level challenges for accessible maps: (i) putting accessibility information in existing maps, and (ii) making existing maps more accessible. As to (i), the major impediment is the lack of data about accessibility in general. As to (ii), a key challenge is to model the user's abilities. And the impediment is that 'ability' is a multifaceted concept. For instance, Froehlich et al. (2019) distinguish between sensory abilities (e.g. vision, hearing), physical abil-

²³There are many more dimensions of spatial querying not discussed here, e.g. objective (e.g. identify, compare), query formulation format (unstructured, formal, visual) and type of user interface feedback (lookup, filtering), see (N. Andrienko, G. Andrienko, and Gatalsky, 2003).

ities (e.g. dexterity, mobility), and cognitive abilities (e.g. memory, concentration, language). 'Expertise' too is a multifaceted concept. There may be at least four aspects to consider: (i) definition of user expertise, (ii) topics of user expertise, (iii) dimensions of user expertise, and (iv) proxy measures for user expertise. Following Downs (2015), an expert can be defined as someone who possesses a certain amount of principled knowledge about a topic. In the case of geovisualization, knowledge is necessary about at least two *topics*: geographic space as well as the operative principles of digital systems. *Dimensions of expertise* may include strategies for problem-solving, the ability to recognize mistakes, and the ability to discriminate between surface (i.e. irrelevant) features and deep (i.e. important) features pertaining to a topic. Finally, *proxy measures* may include experience, performance on specialized tasks, and users' self-assessments. In general, GIScience still lacks means of formally documenting user expertise across evaluation studies of maps and geovisualizations.

Interoperability implies the use of widespread design conventions, that is, the use of design conventions that can be easily recognized by a broad range of users. There are three key challenges here. First, there is evidence that users do not recognize all map symbols to the same extent, but the causes for these discrepancies are not straightforward to pin down. For instance, Horbiński, Cybulski, and Medyńska-Gulij (2020) conducted a study where they investigated the extent to which users associate map symbols used by map providers (e.g. Google Maps, Open Street Maps, Bing Maps, ArcGIS Maps) to their intended function (e.g. search, geolocation, change of layers). Their results suggested that (i) no one provider achieved high recognition rates on all functions tested, and (ii) recognition rates do not necessarily correlate with the frequency of usage of the services. Second, and as mentioned in (Kray, Schmid, and Fritze, 2017; Roth, 2013b), there is still a lack of (empirically-derived) guidelines specific to cartographic interfaces. The third key challenge is that not all design conventions can be encapsulated into tools (à la ColorBrewer) to facilitate their democratization. Examples here include the use (or not use) of north arrows, recommendations about contrast use and symbol grouping to effectively communicate visual hierarchies (Tait, 2018), or effects of the use of visual variables during map icon design (Bell, 2020).

Reuse implies giving users the possibility to transform the visualizations as mentioned above, and necessitates some level of adaptivity on the side of the geovisualization. While adaptivity is a truly desirable feature, realizing it in practice proves challenging because the design space of geovisualizations is complex. It involves at least: the display medium (e.g. mobile phone, large display, desktop computer, tablet computer), interaction modalities (e.g. gaze, speech, touch, mid-air gestures, keyboard, mouse, pen), data attributes (which may be few or many depending on the use case), the measurement scales of the data attributes (e.g. Steven's four levels of measurements), visual variables (up to 12, see Roth, 2017), and tasks (using Brehmer and Munzner (2013)'s typology, the *lower* bound for these is 242).

5.3. The developer's constraints

Findability necessitates the use of aliases to mitigate the vocabulary problem. Previous work reported that 'more is not always better for query expansion' (Degbelo and Teka, 2019). Thus, the key challenge here is to find the optimal aliasing strategy, i.e. the one that will maximize user relevance ratings. In the context of data search, aliasing is often performed on the description of the item at hand (e.g. description metadata field about a given dataset). However, in the case of geovisualizations, content is more than

mere topical description. Content also covers the statements that a user can extract from a geovisualization by looking at it (Scheider, J. Jones, et al., 2014). Developing a proper aliasing strategy for geovisualization search is thus an uncharted research area.

Realizing accessibility from the developer's viewpoint necessitates platform- and device independence (in addition to coping with the diversity of users). Strong constraints are imposed by the device properties (e.g. size, input modalities). In addition, and from the perspective of programming, there is still a need for languages that support plasticity of GI interfaces, so that developers can 'build once, deploy anywhere'. These languages should support both declarative (e.g. a map has two layers), and imperative statements (e.g. re-scale map canvas if device size decreases/increases).

In line with Bishr (1998), two geovisualizations *Geoviz1* and *Geoviz2* are said to be interoperable if Geoviz1 can send requests for services R to Geoviz2, which then returns responses S to Geoviz1, based on a mutual understanding M. The exchange of messages is thus key to realize interoperable geovisualizations. Unique challenges here include a large number of standards for developers to master, and the specification of communication protocols that automatically assess 'mutual understanding' of content to the exchange. As Brodaric (2018) indicated, two types of messages are important in the context of interoperability: control messages informing about how the entities interoperate (e.g. error messages, data requests) and content messages containing actual subject matter. HTTP provides a good starting point for the exchange of *control* messages on the Web (e.g. through the use of HTTP headers and HTTP content negotiation). Knuth, Waitelonis, and Sack (2016) suggested a vision (and preliminary implementation) of using web technologies to exchange rich *content* messages for web media. Their approach relies on the use of HTTP 303 See Other (for redirection to semantic descriptions of media), and the use of RDF as a content description language. Nonetheless, there is still a lack of infrastructure where software agents can autonomously establish the existence of a 'mutual understanding' (e.g. our concepts of 'River' overlap or are equivalent).

Code reuse is an active research topic in the field of software engineering (see e.g. Capilla et al., 2019; Krueger, 1992), and can happen opportunistically (i.e. copypasting snippets from existing programs into new programs) or systematically, with the latter more challenging to realize in practice (see Schmidt, 1999). As discussed in (Mikkonen and Taivalsaari, 2019), the emergence of Internet-based developer forums (e.g. Stack Overflow) and open-source software repositories (e.g. GitHub) has catalyzed opportunistic reuse. Nonetheless, the key challenge here is to define/anticipate reuse units, in order to encapsulate them as modules. As Brereton, Budgen, and Hamilton (1998) indicated: "Web page developers rarely want to reuse existing pages completely, but they may want to reuse existing page components". This likely holds for geovisualizations. Thus, understanding why developers of geovisualizations engage in reuse enterprises, what they expect (i.e. at which granularity reuse happens more often), and issues they face, is a challenge of geovisualization research. This understanding will help design best practices (relevant for instance to the GIScience teaching curriculum) about 'designing for reuse'.

5.4. Summary: what are unique challenges of FAIR geovisualizations?

This section has looked closely into the unique challenges of realizing FAIR geovisualizations. As the section indicated, realizing FAIR geovisualizations demands more of current research regarding provenance description, relevance modelling, user modelling, model-driven development, semantic interoperability, persistence management, licensing policies management and software reuse. As mentioned in Section 2.1, FAIRness is not an absolute state, but a continuum of behaviours (see also, de Miranda Azevedo and Dumontier, 2020). That is, "absolute FAIRness" is not achievable, but online geovisualizations can be made more FAIR, gradually and continually. Also, the challenges listed in this section do not exhibit the same benefit-cost ratio at the time of writing. Addressing the findability of online geovisualizations (steps 1 and 2 of the scenario) seems to offer low-hanging fruits at the moment: the number of online geovisualizations is increasing rapidly and building tools to produce metadata about these so that web-crawlers can semi-automate their retrieval is a task within reach²⁴. If in addition, the metadata is formalized and include machine-readable license information, the computer's demands (Table 1) can already be satisfied to some extent. Accessibility, Interoperability and Reuse challenges from the analyst and developer's viewpoint would need more sustained effort for substantial progress to become visible.

6. The road ahead

Some of the features mentioned in the scenario above (Figure 1) are partly addressed by existing tools. In particular, Hu et al. (2015a) and Mai et al. (2020)'s work on semantic search for ArcGIS online can provide valuable insights as one moves from search of geovisualizations in one geoportal to their search on the Web as a whole. Yet, they are many open questions raised by the scenario. They are summarized in Table 3.

6.1. Findable geovisualizations

Realizing step 1 of the scenario needs work on 'description', i.e. semantic annotation of existing online geovisualizations. Linked Data (Kuhn, Kauppinen, and Janowicz, 2014) has been the de facto technique for semantic annotation of georeferenced data in recent years. Semantic annotation research needs mechanisms to produce largescale annotations for existing geovisualizations on the Web. These annotations need to cope with the peculiarities of geovisualizations, i.e. annotations can be graphical. or textual or a mix of both (see Vanhulst et al., 2018). Also, these annotations need to be fine-grained, that is, capture specifics of the data views (e.g. map, data table, bar chart) composing a geovisualization. Using form-based interaction to produce annotations could be daunting, and we could explore techniques that take advantage of recent progress in speech-based interaction and named entity recognition to enable users to accurately describe cartographic products on the Web. The descriptions should not only touch on missing metadata (e.g. author, creation date), but also inform about content-related aspects of the geovisualizations (e.g. what a user U learned from a geovisualization G during an interaction session I). Work in this area could capitalize on the descriptions of the sub-components (e.g. datasets or software modules/functions/libraries used to produce the geovisualization), if available. Besides, these annotation mechanisms should not be disconnected from existing platforms (e.g. ESRI Storytelling, DataUSA, or the European Open Data Portal), but available 'where the action is', that is, as easy-to-use add-ons within these platforms.

 $^{^{24}}$ See (Lai and Degbelo, 2021) for a recent example of work reporting promising results on semantic metadata generation for online maps.

	Questions		A	D
F	-			
	How to generate large-scale semantic annotations for web-based geovi- sualizations?		х	
	How to enable fine-grained annotation of geovisualization components?		х	
	How to record and share geographic insights across interaction sessions?	x	x	
	How to model relevance and incorporate relevance feedback into geovi- sualization search?		х	x
	How to realize effective aliasing for geovisualization search?			х
	How to design mobile search for geovisualizations?		х	х
	What are developer needs and issues for container-based geovisualization development?			х
Α				
	How to collect and integrate accessibility information in existing geovi- sualizations?	х		х
	How to model user abilities and experience, and include these aspects into relevance rankings?		x	x
	How to automatically verify the accessibility (level) of web-based geovi-			x
	sualizations?			A
	How can accessibility information be gathered on-the-fly, and integrated into relevance ratings?		х	х
	How to realize plastic geovisualizations?		х	х
	How to encode accessibility information about web-based geovisualiza-	x		х
	tions in a machine readable format?			
Ι				
	To which extent are given symbols/palettes intuitive for users and to which extent do they differ across providers?		х	
	How to formalize content of geovisualizations as machine-readable state- ments?	x		x
	How to formalize user insights as machine-readable statements?	x		x
	How to capture connotations of geovisualizations in a machine-readable format?			х
	How to make map design conventions machine-accessible?	x		x
	How to model data gaps formally?	x		x
	How to design interactive guidelines for geovisualization design?		х	x
	How to establish mutual understanding between web agents during geo- data exchange tasks?	х		x
R.				
	How to automatically infer a geovisualization licensing policy from the	x		x
	licensing policies of its components?			
	How to manage licensing policies of geovisualizations over time?			х
	How do geovisualization reuse approaches perform for different groups of users, and how to describe these approaches formally?		х	
	How to estimate transferability of geovisualization components a priori?			x
	What are developer needs and issues for reuse of geovisualization com-			x
	ponents?			
	How to realize digital rights management and smart contracting for geo- visualizations at a Web scale?			x

Table 3. Examples of open questions on the road towards FAIR geovisualizations. A cross indicates the perspectives for which the question are mostly relevant: C stands for computer and indicates knowledge representation issues; A stands for analyst and suggests user interface design issues; and D stands for developer and indicates issues related to technical considerations of publishing online geovisualizations.

Finally, techniques will be needed to maintain these crowdsourced annotations over time, and preserving the multiplicity of views while ruling out mischief.

'Persistence' is mostly relevant to step 2 of the scenario. A user does not want to click on a geovisualization that points to a "HTTP 404 not found" error. Several platforms (e.g. Zenodo, Spatial Data Hub, PURL²⁵) make it now possible to assign persistent URLs (Uniform Resource Locators) to documents on the Web. In the context of Linked Data, the Web community has proposed PingtheSemanticWeb and the "Web of Data - Link Maintenance Protocol" (Volz et al., 2009) as preliminary ideas for protocols to manage the evolution of links on the Web. Traditional web-geovisualization development produces HTML/CSS/JavaScript code deposited on a server, and then accessed from a client that possesses a rendering engine (i.e. the browser). Evolving from this status-quo will necessitate an increasing adoption of a container-based approach to geovisualization development (coupled perhaps with continuous deployment). In addition, application programming interfaces (API) will play a key role, particularly when datasets are too big to be stored locally. For these cases, a persistent way of retrieving datasets at the API level would need to emerge. That is, a DOI-based approach to dataset retrieval is key to accessible geovisualizations. Platforms such as Dataverse²⁶ and DataCite²⁷ allow the assignment of DOIs to datasets. DataCite enables also retrieval of these datasets according to their DOIs at the API level, and is an example implementation of the DOI-based approach to dataset retrieval. Thus, the challenges of persistent geovisualization ahead are more of social nature (i.e. how to ensure widespread adoption) than technical. Further collaborations between relevant institutions (e.g. OGC, the W3C) will be key in overcoming them.

'Relevance' is also a notion that needs to be put under closer scrutiny. In particular, geographic information retrieval research needs to explore how the ranking of geovisualizations can be implemented to produce cognitively plausible results (e.g. how do factors affecting relevance ranking for geovisualizations differ from those used for ranking websites or datasets?). Finally, most graphical perception studies so far have only considered static geovisualizations, but interactivity should receive due attention on the road towards sensible rankings of findable geovisualizations.

Example questions:

- How to generate large-scale semantic annotations for web-based geovisualizations?
- How to enable fine-grained annotation of geovisualization components (e.g. data views, datasets, projection, mapping libraries)?
- How to record and share geographic insights across interaction sessions?
- How to model the relevance of online geovisualizations for the purpose of semantic search?
- How to realize effective aliasing for geovisualization search?

6.2. Accessible geovisualizations

As mentioned above, modeling user abilities/expertise, and realizing plastic geovisualization presents opportunities for further research on accessible geovisualization, i.e. tailoring to abilities and expertise, platform- and device-independence. There is also a need for tools translate accessibility information into machine readable format,

 $^{^{25}}$ www.purl.org

 $^{^{26}}$ https://dataverse.org/

²⁷https://datacite.org/

wherever possible, so that these can be used during the implementation of geovisualizations. At the moment, there is still a disconnect between work on empirically derived recommendations regarding the accessibility of web maps (e.g. Hennig, Zobl, and Wasserburger, 2017) and work on technical means to document accessibility requirements for web-pages (e.g. Harper and Yesilada, 2007; Pelzetter, 2020).

Example questions:

- How to collect and integrate accessibility information in existing geovisualizations?
- How to model user abilities and experience, and include these aspects into relevance rankings?
- How to automatically verify the accessibility (level) of web-based geovisualizations?
- How can accessibility information be gathered on-the-fly, and integrated into relevance ratings?
- How to realize plastic geovisualizations?
- How to encode accessibility information about web-based geovisualizations in a machine readable format?

6.3. Interoperable geovisualizations

Interoperability of geovisualizations (i.e. formalization of statements, compliance with widespread design conventions) is relevant to steps 2, 3 and 4 of the scenario. There is a need for user testing to evaluate actual user interpretation of map symbols, potential misunderstandings, and the reasons for these. This research will be mostly valuable for step 2. Despite much research on the interpretation of map symbols (e.g. Akella, 2009; Kinkeldev et al., 2014; Korpi and Ahonen-Rainio, 2010; Koylu and Guo, 2017; Schnürer et al., 2020), formal rules regarding map symbolization are still not consolidated, perhaps due to the diversity of map context usages available and studied. Initiatives such as the GIS&T Body of Knowledge (Section: Cartography & Visualization, https://gistbok.ucgis.org/), which synthesize insights from the academic literature and communicate them in a language accessible to non-experts will thus play an important role for awareness-raising, the adoption of best practices, and the realization of more interoperable geovisualizations (analyst's perspective). In addition, advances on semantic interoperability research (e.g. formal specification of geodatasets' content and geovisualization components) is needed to realize steps 3 and 4 of the scenario. So far, semantic interoperability research in GIScience has invested heavily in ontology building. These ontologies have been used to generate theories of geoinformation (e.g. Agarwal, 2005; Bittner, Donnelly, and Smith, 2009; Couclelis, 2010), for query disambiguation and expansion (e.g. C. B. Jones et al., 2004; Lutz and Klien, 2006; Purves, Clough, C. B. Jones, Arampatzis, et al., 2007), and for the (semi-automatic) composition of geoprocessing services (e.g. Lemmens et al., 2006). Ontologies are also key to the Sensor Web Enablement initiative of the OGC (see Bröring et al., 2011). While these works offer valuable insight, *demos of agents* that use these ontologies to automatically exchange things during a communication process have been less frequent. Ergo, GIScience research needs to go beyond ontology building to embrace *semantic* agent building, where the agents select ontologies relevant to a given scenario, and exchange information with other relevant agents in an autonomous way, to solve some specific tasks. This is a major gap in current research. Identifying the tasks that cover scenarios of value in GIScience, and following up with agents that solve these tasks,

will provide a much clearer measure of progress for semantic interoperability research than ontology provision alone. As to coping with missing data, formalizing and providing solutions to the issue has been relatively rare in geovisualization research. Notable exceptions are Robinson (2019)'s typology of strategies to draw users' attention to the presence of gaps in the data, and Ballatore (2019)'s context frame approach to generates visual cues about objects not shown on the map in order to minimize the need for zooming operations during an interaction session.

Moving to the point were semantic agents can seamlessly cooperate, needs advances along three axes: agents, protocols, and messages. The exact role of agents remains to be specified, but at least four roles can be envisioned: discovery (i.e. find out other visualizations on the Web that provide different perspectives on the data at hand); negotiation (e.g. establish with other visualizations on the Web whether or not data transfer would be possible); retrieval (i.e. get pieces of data from other visualizations); and integration (i.e. close data gaps of Geoviz1 with data items from Geoviz 2, see Figure 1). Next to agent-oriented work, protocol-oriented work is needed to provide ways of exchanging messaging, building for example on state-of-the-art web technologies. Finally, message-oriented work can provide a uniform way of formally representing the content of datasets underlying the geovisualizations (e.g. their topic & typical spatiotemporal attributes such as extent, temporal coverage, accuracy, and resolution), and of the geovisualizations themselves. Kuhn (2012)'s core concepts of spatial information may provide a starting point for a vocabulary for data content annotation; Roth (2013a)'s taxonomy of interaction primitives for interactive geovisualization may also be useful for the annotation of geovisualizations. It remains to be seen whether these existing vocabularies/taxonomies are sufficient, or whether they should be extended (and in which way) to enable message exchanging for interoperable geovisualizations. Finally, a measure of more social nature on the roads towards interoperable geovisualizations could be the organization of (yearly) contests, ending up with demos showing geovisualizations exchanging some data autonomously (or deciding not to exchange data, based on a mutual understanding). Plaisant, Fekete, and Grinstein (2008) reported positive experiences from annual competitions for the development of new visualization systems in the Information Visualization community.

Example questions:

- To which extent are given symbols/palettes intuitive for users and to which extent do they differ across providers?
- How to formalize the content of geovisualizations as machine-readable statements?
- How to formalize user insights as machine-readable statements?
- How to capture connotations of geovisualizations in a machine-readable format?
- How to model data gaps formally?
- How to design interactive guidelines²⁸ for geovisualization design?
- How to establish mutual understanding between web agents during geodata exchange tasks?

6.4. Reusable geovisualizations

Reuse (i.e. licensing, transformability, and modularity) is primarily relevant to steps 5 and 6 of the scenario. Concepts such as 'toolkits', 'map use contexts', 'intelligent geo-

 $^{^{28}}$ Interactive guidelines walk users through problem-solution patterns in an interactive manner, to transfer insights from the academic literature to non-experts (Trilles et al., 2020).

visualization', and 'map plasticity' have provided a good start to pinpointing the many dimensions of geovisualization reuse. Yet, we still need a clearer picture of reusability approaches available to geovisualizations creators and users, and most importantly, of the needs of users within each of the approaches. Put differently, a theoretical description of reusability approaches and a comparison of their merits is a challenging, yet much need contribution necessary to advance the current state-of-the-art on reusable geovisualization. As mentioned in Section 4.2, template-based designs realize reuse across topics effectively: the same visualization template (e.g. choropleth, graduated symbol, cartogram) can be used across different topical domains (e.g. politics, demographics, real estate). The idea of template reuse in one-click (e.g. drag and drop) is appealing, but may be challenging to fully realize given the complexity of the design space of geovisualizations. Yet, it could be valuable during the comparison of geovisualization reuse approaches to assess research progress. For example, the number of clicks needed to create a visualization can be used along with established measures such as expressiveness (i.e. the number of geovisualization types that can be created), during the evaluation of tools and toolkits facilitating geovisualization reuse. An early example of a drag-and-drop approach to visualize datasets was proposed in (Wills and L. Wilkinson, 2010). The approach focused on detecting structural (as opposed to semantical) aspects of datasets, relies on the grammar of graphics, and was illustrated primarily on non-geographic datasets. Extending it to account at least for semantical and spatial aspects of datasets would be needed to fully realize steps 5 and 6 of the scenario. Finally, a more social aspect of geovisualization reuse touches upon the licensing policy. By default, nothing is reusable unless permission is explicitly granted. This principle does not adequately reflect the state of matters on the Web. To give an example, the source code of web-based geovisualizations can be accessed through browser inspection tools. If the visualization has been developed with a library based on Web standards such as HTML, CSS (e.g. D3.js), data items powering the visualization can also be accessed. That is, licensing of web-based geovisualization may consider moving to a new motto: 'everything is reusable unless reuse is explicitly restricted' to reflect openness of resources on the Web.

Example questions:

- How to automatically infer a geovisualization's licensing policy from the licensing policies of its components?
- How to manage licensing policies of geovisualizations over time?
- How do geovisualization reuse approaches perform for different groups of users, and how to describe these approaches formally?
- How to estimate the transferability of geovisualization components a priori?
- What are developer needs and issues for reuse of geovisualization components?

6.5. Summary: what are open questions for FAIR geovisualizations research?

Revisiting the scenario introduced at the beginning of the article, this section has contributed open questions on the road towards FAIR geovisualizations. The questions listed are by no means exhaustive, but illustrate that new types of problems arise to realize findability, access, interoperability, and reuse of web-based geovisualizations. Immediate, practical steps to make progress on steps 1-3 of the scenario could include the use of knowledge harvesting techniques (Weikum, Hoffart, and Suchanek, 2019) and of semantic annotation to generate knowledge graphs from geovisualizations stored as



Figure 4. An example of practical step towards more FAIR geovisualizations. Top: current online visualizations are treated as html documents for search; Bottom: exploring how knowledge harvesting techniques and semantic annotation can help build knowledge graphs for advanced search of geovisualizations at a Web scale.

HTML-documents, the development of techniques to search for relevant geovisualizations over these knowledge graphs, and techniques to maintain online geovisualizations and their respective knowledge graphs synchronized (Figure 4). Using visualizations of the Observable gallery as a testbed could be helpful to start, as these are already annotated using the Open Graph Protocol²⁹.

7. Limitations

This article should be thought of as an initial exploration of FAIR geovisualizations, not the final word. There are some limitations that can be noted.

First, the work has argued that geovisualizations should be considered as a unit in their own right, for geographic information search and reuse on the Web. Getting there necessitates more work on making geovisualizations fair (in the literal sense) to the datasets they depict. There are indeed numerous factors that affect visual encodings that may impede the fairness of visualization with respect to the data they portray. These include, among others, graphical inference (are the patterns we see really there?, see Wickham et al., 2010), Tufte lie factor (is the size of the effect shown in the graphic directly proportional to the size of the effect in the data), perceptual effectiveness (is the visual symbol to encode differences appropriately chosen?), and imperfections of data classification (Monmonier, 2005). These issues, though acknowledged, are not discussed in detail here. Accounting for errors occurring in data-to-visuals transformation processes, and more broadly of uncertainties inherent to data exploration processes and their visual communication, can be examined once prototypical applications emerge that realize the steps presented in Figure 1.

Second, as to the formulation of the open questions, GIScience has often used fore-

 $^{^{29}\}mathrm{Thanks}$ to one anonymous reviewer for pointing at Observable.

casting to articulate research agendas. For example, forecasting has been used to predict the evolution of spatial data infrastructures in (Diaz et al., 2012), anticipate the development of the digital earth in (Craglia et al., 2012; Goodchild et al., 2012), and conjecture developments of the field as a whole in (Goodchild, 2009, 2010). We've learned that forecasting may miss important developments (for an anecdote, see Goodchild, 2009), and backcasting too may miss important developments of the years to come. Finally, similar to forecasting, backcasting is 'value-driven' (Dreborg, 1996), that is, the ideas are inevitably biased towards the author's research.

8. Conclusion

This work has applied the FAIR principles to a new domain (i.e. geovisualization). and taken a holistic approach to FAIRness to uncover some challenges and opportunities for GIScience research. This article contributes to geovisualization research in several ways. First, the work highlighted that geovisualizations are information products increasingly available, but sufficiently distinct from websites and raw datasets to deserve sustained effort aiming at making them FAIR (Findable, Accessible, Interoperable and Reusable). Second, the work discussed three complementary perspectives on FAIR geovisualizations (the computer, the analyst, and the developer), and clarified their respective demands. The framework resulting from the discussion can help researchers involved in realizing FAIR geovisualization to disambiguate their stance. It can also be used to systematically map research on FAIR geovisualizations as it evolves. Third, relevant approaches from the literature to realize FAIRness and some unique challenges of FAIR geovisualizations were brought forth and discussed. At last. the article sketched open questions towards FAIR geovisualizations. These questions arguably present huge challenges for current research but challenges worth attending to. The future of geovisualization research and practice is FAIR.

Acknowledgements

I thank several anonymous reviewers for their detailed and very insightful comments on earlier versions of this article. The icons in Figures 1 and 4 were taken from www. flaticon.com (designers: 'Freepik', 'Kiranshastry', 'Nhor Phai', 'Pixel perfect', and 'srip').

Supplementary material

The supplementary material is available at https://doi.org/10.6084/m9. figshare.13635563.

References

Aditya, Trias and Menno-Jan Kraak (June 2007). "A search interface for an SDI: implementation and evaluation of metadata visualization strategies". In: *Transactions in GIS* 11.3, pp. 413–435. DOI: 10.1111/j.1467-9671.2007.01053.x.

- Agarwal, P (2005). "Ontological considerations in GIScience". In: International Journal of Geographical Information Science 19.5, pp. 501–536. DOI: 10.1080/ 13658810500032321.
- Akella, Mamata Kumari (Jan. 2009). "First responders and crisis map symbols: clarifying communication". In: Cartography and Geographic Information Science 36.1, pp. 19–28. DOI: 10.1559/152304009787340179.
- Andrienko, Gennady, Natalia Andrienko, Urska Demsar, et al. (2010). "Space, time and visual analytics". In: International Journal of Geographical Information Science 24.10, pp. 1577–1600. DOI: 10.1080/13658816.2010.508043.
- Andrienko, Gennady, Natalia Andrienko, Piotr Jankowski, et al. (Sept. 2007). "Geovisual analytics for spatial decision support: Setting the research agenda". In: International Journal of Geographical Information Science 21.8, pp. 839–857. DOI: 10.1080/13658810701349011.
- Andrienko, Natalia and Gennady Andrienko (Oct. 2008). "Evacuation trace mini challenge award: Tool integration analysis of movements with Geospatial Visual Analytics Toolkit". In: VAST'08 IEEE Symposium on Visual Analytics Science and Technology. Columbus, Ohio, USA: IEEE, pp. 205–206. DOI: 10.1109/VAST.2008. 4677388.
- Andrienko, Natalia, Gennady Andrienko, and Peter Gatalsky (Dec. 2003). "Exploratory spatio-temporal visualization: an analytical review". In: Journal of Visual Languages & Computing 14.6, pp. 503–541. DOI: 10.1016/S1045-926X(03)00046-6.
- Avram, Henriette D. (2003). "Machine-readable cataloging (MARC) program". In: Encyclopedia of library and information science 3, pp. 1712–1730. DOI: 10.1081/E-ELIS120008993.
- Badam, Sriram Karthik and Niklas Elmqvist (Jan. 2019). "Visfer: camera-based visual data transfer for cross-device visualization". In: *Information Visualization* 18.1, pp. 68–93. DOI: 10.1177/1473871617725907.
- Ballatore, Andrea (2019). "A context frame for interactive maps". In: Accepted Short Papers and Posters from the 22nd AGILE Conference on Geo-information Science (AGILE 2019). Ed. by Phaedon Kyriakidis et al. Cyprus, Greece.
- Battle, Leilani et al. (2018). "Beagle: automated extraction and interpretation of visualizations from the Web". In: *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems (CHI 2018)*. Ed. by Regan L Mandryk et al. Montreal, Quebec, Canada: ACM, p. 594. DOI: 10.1145/3173574.3174168.
- Bell, Sarah (Jan. 2020). "Map icon design". In: Geographic Information Science & Technology Body of Knowledge. Ed. by John P. Wilson. Q1. DOI: 10.22224/gistbok/2020.1.6.
- Bertin, Jacques (1983). Semiology of graphics: diagrams, networks, maps (Translated by William J. Berg). The University of Wisconsin Press, Madison.
- Bibri, Simon Elias (Dec. 2018). "Backcasting in futures studies: a synthesized scholarly and planning approach to strategic smart sustainable city development". In: *European Journal of Futures Research* 6.1, p. 13. DOI: 10.1186/s40309-018-0142-z.
- Bishr, Yaser (June 1998). "Overcoming the semantic and other barriers to GIS interoperability". In: International Journal of Geographical Information Science 12.4, pp. 299–314. DOI: 10.1080/136588198241806.
- Bittner, Thomas, Maureen Donnelly, and Barry Smith (2009). "A spatio-temporal ontology for geographic information integration". In: *International Journal of Geographical Information Science* 23.6, pp. 765–798. DOI: 10.1080 / 13658810701776767.

- Bostock, Michael and Jason Davies (May 2013). "Code as cartography". In: *The Cartographic Journal* 50.2, pp. 129–135. DOI: 10.1179/0008704113Z.0000000078.
- Bostock, Michael and Jeffrey Heer (Nov. 2009). "Protovis: A graphical toolkit for visualization". In: *IEEE Transactions on Visualization and Computer Graphics* 15.6, pp. 1121–1128. DOI: 10.1109/TVCG.2009.174.
- Bostock, Michael, Vadim Ogievetsky, and Jeffrey Heer (Dec. 2011). "D³ Data-Driven Documents". In: *IEEE Transactions on Visualization and Computer Graphics* 17.12, pp. 2301–2309. DOI: 10.1109/TVCG.2011.185.
- Brehmer, Matthew and Tamara Munzner (Dec. 2013). "A multi-level typology of abstract visualization tasks". In: *IEEE Transactions on Visualization and Computer Graphics* 19.12, pp. 2376–2385. DOI: 10.1109/TVCG.2013.124.
- Brereton, P., D. Budgen, and G. Hamilton (1998). "Hypertext: the next maintenance mountain". In: *Computer* 31.12, pp. 49–55. DOI: 10.1109/2.735850.
- Brewer, Cynthia A. (Jan. 2003). "A transition in improving maps: the ColorBrewer example". In: *Cartography and Geographic Information Science* 30.2, pp. 159–162. DOI: 10.1559/152304003100011126.
- Brodaric, Boyan (Mar. 2018). "Interoperability of representations". In: International Encyclopedia of Geography. Ed. by Douglas Richardson et al. Oxford, UK: John Wiley & Sons, Ltd, pp. 1–18. DOI: 10.1002/9781118786352.wbieg0894.pub2.
- Bröring, Arne et al. (Mar. 2011). "New Generation Sensor Web Enablement". In: Sensors 11.3, pp. 2652–2699. DOI: 10.3390/s110302652.
- Brovelli, Maria A. et al. (Apr. 2012). "Web geoservices and ancient cadastral maps: The Web C.A.R.T.E. Project". In: *Transactions in GIS* 16.2, pp. 125–142. DOI: 10.1111/j.1467-9671.2012.01311.x.
- Capilla, Rafael et al. (Aug. 2019). "Opportunities for software reuse in an uncertain world: From past to emerging trends". In: *Journal of Software: Evolution and Process* 31.8. DOI: 10.1002/smr.2217.
- Cardoso, Nuno and Mário J Silva (2007). "Query expansion through geographical feature types". In: Proceedings of the 4th ACM Workshop On Geographic Information Retrieval (GIR 2007). Ed. by Ross Purves and Christopher B Jones. Lisbon, Portugal: ACM, pp. 55–60. DOI: 10.1145/1316948.1316963.
- Chiang, Yao-Yi, Weiwei Duan, et al. (2020). "Creating structured, linked geographic data from historical maps: challenges and trends". In: Using Historical Maps in Scientific Studies. Ed. by Yao-Yi Chiang et al. Springer, Cham, pp. 37–63. DOI: 10.1007/978-3-319-66908-3_3.
- Chiang, Yao-Yi, Stefan Leyk, and Craig A. Knoblock (May 2014). "A survey of digital map processing techniques". In: *ACM Computing Surveys* 47.1, pp. 1–44. DOI: 10. 1145/2557423.
- Cleveland, William S. and Robert McGill (Sept. 1984). "Graphical perception: theory, experimentation, and application to the development of graphical methods". In: *Journal of the American Statistical Association* 79.387, p. 531. DOI: 10.2307/2288400.
- Çöltekin, Arzu, Susanne Bleisch, et al. (Oct. 2017). "Persistent challenges in geovisualization a community perspective". In: *International Journal of Cartography* 3.sup1, pp. 115–139. DOI: 10.1080/23729333.2017.1302910.
- Çöltekin, Arzu, Benedikt Heil, et al. (Jan. 2009). "Evaluating the effectiveness of interactive map interface designs: A case study integrating usability metrics with eye-movement analysis". In: Cartography and Geographic Information Science 36.1, pp. 5–17. DOI: 10.1559/152304009787340197.

- Çöltekin, Arzu, Halldor Janetzko, and Sara Irina Fabrikant (2018). "Geovisualization".
 In: The Geographic Information Science & Technology Body of Knowledge (2nd Quarter 2018 Edition). Ed. by John P. Wilson. DOI: 10.22224/gistbok/2018.2.6.
- Couclelis, Helen (2010). "Ontologies of geographic information". In: International Journal of Geographical Information Science 24.12, pp. 1785–1809. DOI: 10.1080/ 13658816.2010.484392.
- Craglia, Max et al. (Jan. 2012). "Digital Earth 2020: towards the vision for the next decade". In: *International Journal of Digital Earth* 5.1, pp. 4–21. DOI: 10.1080/17538947.2011.638500.
- de Miranda Azevedo, Ricardo and Michel Dumontier (Jan. 2020). "Considerations for the conduction and interpretation of FAIRness evaluations". In: Data Intelligence 2.1-2, pp. 285-292. ISSN: 2641-435X. DOI: 10.1162/dint_a_00051. URL: https: //direct.mit.edu/dint/article/2/1-2/285-292/10015.
- Degbelo, Auriol (2017). "Linked data and visualization: two sides of the transparency coin". In: *Proceedings of the 3rd ACM SIGSPATIAL Workshop on Smart Cities and Urban Analytics UrbanGIS'17*. Ed. by Huy T. Vo and Bill Howe. Los Angeles, California, USA: ACM Press, pp. 1–8. DOI: 10.1145/3152178.3152191.
- (2020a). "An ontology design pattern for geovisualization content description". In: *Advances in Pattern-based Ontology Engineering*. Ed. by Eva Blomqvist et al. IOS Press.
- (Apr. 2020b). "Open data user needs: a preliminary synthesis". In: Companion Proceedings of the Web Conference 2020. Ed. by Amal El Fallah Seghrouchni et al. Taipei, Taiwan: ACM, pp. 834–839. DOI: 10.1145/3366424.3386586.
- Degbelo, Auriol, Carlos Granell, et al. (2016). "Opening up smart cities: citizen-centric challenges and opportunities from GIScience". In: *ISPRS International Journal of Geo-Information* 5.2, p. 16. DOI: 10.3390/ijgi5020016.
- Degbelo, Auriol and Tomi Kauppinen (2018). "Increasing transparency through web maps". In: Companion of Proceedings of the Web Conference 2018 WWW '18.
 Ed. by Pierre-Antoine Champin et al. Lyon, France: ACM Press, pp. 899–904. DOI: 10.1145/3184558.3191515.
- Degbelo, Auriol and Christian Kray (2018). "Intelligent geovisualizations for open government data (vision paper)". In: 26th ACM SIGSPATIAL International Conference on Advances in Geographic Information Systems. Ed. by Farnoush Banaei-Kashani et al. Seattle, Washington, USA: ACM Press, pp. 77–80. DOI: 10.1145/ 3274895.3274940.
- Degbelo, Auriol, Saad Sarfraz, and Christian Kray (Mar. 2020). "Data scale as Cartography: a semi-automatic approach for thematic web map creation". In: *Cartography* and *Geographic Information Science* 47.2, pp. 153–170. DOI: 10.1080/15230406. 2019.1677176.
- Degbelo, Auriol and Ang Sherpa (Apr. 2020). "Open geodata reuse: towards natural language interfaces to web APIs". In: *Companion Proceedings of the Web Conference 2020.* Ed. by Amal El Fallah Seghrouchni et al. Taipei, Taiwan: ACM, pp. 703–710. DOI: 10.1145/3366424.3384363.
- Degbelo, Auriol and Brhane Teka (2019). "Spatial search strategies for open government data: A systematic comparison". In: 13th Workshop on Geographic Information Retrieval (GIR'19). Ed. by Ross Purves and Christopher B Jones. Lyon, France: ACM Press, 2:1–2:10. DOI: 10.1145/3371140.3371142.
- Degbelo, Auriol, Sergio Trilles, et al. (2016). "Designing semantic application programming interfaces for open government data". In: *JeDEM - eJournal of eDemocracy* and Open Government 8.2, pp. 21–58. DOI: 10.29379/jedem.v8i2.420.

- Degbelo, Auriol, Jonas Wissing, and Tomi Kauppinen (Oct. 2018). "A comparison of geovisualizations and data tables for transparency enablement in the open government data landscape". In: *International Journal of Electronic Government Research* 14.4, pp. 39–64. DOI: 10.4018/IJEGR.2018100104.
- Diaz, Laura et al. (2012). "Future SDI Impulses from Geoinformatics research and IT trends". In: International Journal of Spatial Data Infrastructures Research 7, pp. 378–410. DOI: 10.2902/1725-0463.2012.07.art18.
- Döllner, Jürgen (2005). "Geospatial digital rights management in geovisualization". In: *The Cartographic Journal* 42.1, pp. 27–34. DOI: 10.1179/000870405X57266.
- Downs, Roger M (2015). "Exploring the nature and development of expertise in Geography". In: Space in Mind: Concepts for Spatial Learning and Education. Ed. by Daniel Montello, Karl Grossner, and Donald Janelle. MIT press. Chap. 14, pp. 281– 305.
- Dreborg, Karl H (Nov. 1996). "Essence of backcasting". In: *Futures* 28.9, pp. 813–828. DOI: 10.1016/S0016-3287(96)00044-4.
- Fechner, Thore and Christian Kray (2014). "Georeferenced open data and augmented interactive geo-visualizations as catalysts for citizen engagement". In: *eJournal of eDemocracy and Open Government* 6.1, pp. 14–35. DOI: 10.29379/jedem.v6i1. 283.
- Fechner, Thore, Dennis Wilhelm, and Christian Kray (2015). "Ethermap real-time collaborative map editing". In: Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems - CHI '15. Ed. by Bo Begole et al. Seoul, South Korea: ACM Press, pp. 3583–3592. DOI: 10.1145/2702123.2702536.
- Feitelson, Dror et al. (2020). "How developers choose names". In: *IEEE Transactions* on Software Engineering, pp. 1–1. DOI: 10.1109/TSE.2020.2976920.
- Froehlich, Jon E. et al. (Feb. 2019). "Grand challenges in accessible maps". In: Interactions 26.2, pp. 78–81. DOI: 10.1145/3301657.
- Fu, Gaihua, Christopher B Jones, and Alia I Abdelmoty (2005). "Ontology-based spatial query expansion in information retrieval". In: On the Move to Meaningful Internet Systems 2005. Ed. by Robert Meersman et al. Vol. 3761. Springer, pp. 1466– 1482. DOI: 10.1007/11575801_33.
- Furnas, George W (1985). "Experience with an adaptive indexing scheme". In: Proceedings of the SIGCHI conference on Human factors in computing systems CHI '85. Ed. by Lorraine Borman and Raoul Smith. New York, New York, USA: ACM Press, pp. 131–135. DOI: 10.1145/317456.317480.
- Furnas, George W et al. (Nov. 1987). "The vocabulary problem in human-system communication". In: Communications of the ACM 30.11, pp. 964–971. DOI: 10. 1145/32206.32212.
- Gao, Song, Krzysztof Janowicz, and Dingwen Zhang (2016). "Designing a map legend ontology for searching map content". In: Advances in Ontology Design and Patterns [revised and extended versions of the papers presented at the 7th edition of the Workshop on Ontology and Semantic Web Patterns (WOP@ISWC 2016). Ed. by Karl Hammar et al. Vol. 32. IOS Press, pp. 119–130. DOI: 10.3233/978-1-61499-826-6-119.
- Goodchild, M. F. (Nov. 2009). "Geographic information systems and science: today and tomorrow". In: Annals of GIS 15.1, pp. 3–9. DOI: 10.1080 / 19475680903250715.
- (2010). "Twenty years of progress: GIScience in 2010". In: Journal of Spatial Information Science 1, pp. 3–20.

- Goodchild, M. F. et al. (July 2012). "Next-generation digital earth". In: *Proceedings* of the National Academy of Sciences 109.28, pp. 11088–11094. DOI: 10.1073/pnas. 1202383109.
- Gould, Nicholas and William Mackaness (May 2016). "From taxonomies to ontologies: formalizing generalization knowledge for on-demand mapping". In: *Cartography and Geographic Information Science* 43.3, pp. 208–222. DOI: 10.1080/15230406.2015. 1072737.
- Grammel, Lars et al. (2013). "A survey of visualization construction user interfaces". In: *EuroVis - Short Papers*. Ed. by Mario Hlawitschka and Tino Weinkauf. Leipzig, Germany: The Eurographics Association. DOI: 10.2312/PE.EuroVisShort. EuroVisShort2013.019-023.
- Graves, Alvaro and James Hendler (2013). "Visualization tools for open government data". In: *Proceedings of the 14th Annual International Conference on Digital Government Research dg.o '13.* Ed. by Sehl Mellouli, Luis F. Luna-Reyes, and Jing Zhang. Quebec, Canada: ACM Press, pp. 136–145. DOI: 10.1145/2479724. 2479746.
- (2014). "A study on the use of visualizations for Open Government Data". In: Information Polity 19.1,2, pp. 73–91. DOI: 10.3233/IP-140333.
- Griffin, Amy L. (June 2020). "Trustworthy maps". In: Journal of Spatial Information Science 20. DOI: 10.5311/JOSIS.2020.20.654.
- Griffin, Amy L. et al. (Oct. 2017). "Designing across map use contexts: a research agenda". In: *International Journal of Cartography* 3.sup1, pp. 90–114. DOI: 10. 1080/23729333.2017.1315988.
- Guidero, Elaine (July 2017). "Typography". In: Geographic Information Science & Technology Body of Knowledge. Ed. by John P. Wilson. Q3. DOI: 10.22224/gistbok/2017.3.2.
- Hardisty, Frank and Anthony C. Robinson (Mar. 2011). "The GeoViz toolkit: using component-oriented coordination methods for geographic visualization and analysis". In: International Journal of Geographical Information Science 25.2, pp. 191– 210. DOI: 10.1080/13658810903214203.
- Harper, Simon and Yeliz Yesilada (Sept. 2007). "Web Authoring for Accessibility (WAfA)". In: Web Semantics: Science, Services and Agents on the World Wide Web 5.3, pp. 175–179. DOI: 10.1016/j.websem.2007.05.001.
- Harrie, Lars, Hanna Stigmar, and Milan Djordjevic (Mar. 2015). "Analytical estimation of map readability". In: *ISPRS International Journal of Geo-Information* 4.2, pp. 418–446. DOI: 10.3390/ijgi4020418.
- Harrower, Mark and Cynthia A. Brewer (June 2003). "ColorBrewer.org: An online tool for selecting colour schemes for maps". In: *The Cartographic Journal* 40.1, pp. 27– 37. DOI: 10.1179/000870403235002042.
- Haslauer, Eva, Markus Biberacher, and Thomas Blaschke (May 2016). "A spatially explicit backcasting approach for sustainable land-use planning". In: *Journal of Environmental Planning and Management* 59.5, pp. 866–890. ISSN: 0964-0568. DOI: 10.1080/09640568.2015.1044652.
- Hennig, Sabine, Fritz Zobl, and Wolfgang W. Wasserburger (Nov. 2017). "Accessible web maps for visually impaired users: recommendations and example solutions". In: *Cartographic Perspectives* 88, pp. 6–27. DOI: 10.14714/CP88.1391.
- Horbiński, Tymoteusz, Paweł Cybulski, and Beata Medyńska-Gulij (2020). "Graphic design and button placement for mobile map applications". In: *The Cartographic Journal* 0.0, pp. 1–13. DOI: 10.1080/00087041.2019.1631008.

- Hu, Yingjie et al. (2015a). "Enabling semantic search and knowledge discovery for ArcGIS online: A linked-data-driven approach". In: AGILE 2015 - Geographic Information Science as an Enabler of Smarter Cities and Communities. Ed. by Fernando Bacao, Maribel Yasmina Santos, and Marco Painho. Lisbon, Portugal: Springer, Cham, pp. 107–124. DOI: 10.1007/978-3-319-16787-9_7.
- (June 2015b). "Metadata topic harmonization and semantic search for linked-datadriven geoportals: A case study using ArcGIS online". In: *Transactions in GIS* 19.3, pp. 398–416. DOI: 10.1111/tgis.12151.
- Huang, Weiming and Lars Harrie (Apr. 2019). "Towards knowledge-based geovisualisation using Semantic Web technologies: a knowledge representation approach coupling ontologies and rules". In: *International Journal of Digital Earth*, pp. 1–22. DOI: 10.1080/17538947.2019.1604835.
- Ipatow, Natalia and Francis Harvey (Dec. 2019). "Bertin's graphic variables and online map makers: An empirical study of maps produced by prosumers and cartographers". In: Cartographica: The International Journal for Geographic Information and Geovisualization 54.4, pp. 233–244. DOI: 10.3138/cart.54.4.2018-0024.
- Jacobsen, Annika et al. (Jan. 2020). "FAIR principles: interpretations and implementation considerations". In: *Data Intelligence* 2.1-2, pp. 10–29. DOI: 10.1162/dint_ r_00024.
- Janowicz, Krzysztof et al. (2010). "Semantic enablement for spatial data infrastructures". In: *Transactions in GIS* 14.2, pp. 111–129. DOI: 10.1111/j.1467-9671. 2010.01186.x.
- Javed, Waqas, Bryan McDonnel, and Niklas Elmqvist (Nov. 2010). "Graphical perception of multiple time series". In: *IEEE Transactions on Visualization and Computer Graphics* 16.6, pp. 927–934. ISSN: 1077-2626. DOI: 10.1109/TVCG.2010.162.
- Jiang, Yongyao et al. (Sept. 2018). "Towards intelligent geospatial data discovery: a machine learning framework for search ranking". In: International Journal of Digital Earth 11.9, pp. 956–971. DOI: 10.1080/17538947.2017.1371255.
- Jones, Christopher B et al. (2004). "The SPIRIT spatial search engine: architecture, ontologies and spatial indexing". In: *Geographic Information Science*. Ed. by Max J Egenhofer, Christian Freksa, and Harvey J Miller. Adelphi, Maryland, USA: Springer Berlin Heidelberg, pp. 125–139.
- Jones, Jim et al. (2014). "Making the web of data available via web feature services".
 In: The 17th AGILE International Conference on Geographic Information Science
 Connecting a Digital Europe Through Location and Place. Ed. by J Huerta, S Schade, and C Granell. Castellón, Spain: Springer International Publishing, pp. 341–361.
- Käfer, Tobias et al. (2013). "Observing linked data dynamics". In: The Semantic Web: Semantics and Big Data, 10th International Conference (ESWC 2013). Ed. by Philipp Cimiano et al. Montpellier, France: Springer Berlin, pp. 213–227. DOI: 10.1007/978-3-642-38288-8_15.
- Kent, Robert B. and Richard E. Klosterman (June 2000). "GIS and mapping". In: Journal of the American Planning Association 66.2, pp. 189–198. DOI: 10.1080/ 01944360008976098.
- Kinkeldey, Christoph et al. (Oct. 2014). "Evaluation of noise annotation lines: using noise to represent thematic uncertainty in maps". In: Cartography and Geographic Information Science 41.5, pp. 430–439. DOI: 10.1080/15230406.2014.949868.
- Kiser, Tim and Nicole Smeltekop (Sept. 2018). "A method for creating scanned map metadata for geoportals, library catalogs, and digital repositories: reworking existing MARC records of paper maps to create new records for their scanned counterparts".

In: Journal of Map & Geography Libraries 14.2-3, pp. 109–131. DOI: 10.1080/ 15420353.2019.1640166.

- Knuth, Magnus, Jörg Waitelonis, and Harald Sack (2016). "I am a machine, let me understand web media!" In: International Conference on Web Engineering (ICWE 2016). Ed. by Alessandro Bozzon, Philippe Cudre-Maroux, and Cesare Pautasso. Lugano, Switzerland: Springer International Publishing, pp. 467–475. DOI: 10.1007/978-3-319-38791-8_33.
- Köbben, B.J. and Menno-Jan Kraak (2020). "Web mapping". In: International Encyclopedia of Human Geography. Ed. by Audrey Kobayashi. Elsevier, pp. 333–337. DOI: 10.1016/B978-0-08-102295-5.10565-7.
- Koehler, Wallace (1999). "An analysis of web page and web site constancy and permanence". In: Journal of the American Society for Information Science 50.2, pp. 162–180. DOI: 10.1002/(SICI)1097-4571(1999)50:2<162::AID-ASI7>3.0.CO;2-B.
- Korpi, Jari and Paula Ahonen-Rainio (Nov. 2010). "Cultural constraints in the design of pictographic symbols". In: *The Cartographic Journal* 47.4, pp. 351–359. DOI: 10.1179/000870410X12911337964923.
- Koylu, Caglar and Diansheng Guo (Oct. 2017). "Design and evaluation of line symbolizations for origin-destination flow maps". In: *Information Visualization* 16.4, pp. 309–331. DOI: 10.1177/1473871616681375.
- Kraak, Menno-Jan and Sara Irina Fabrikant (Oct. 2017). "Of maps, cartography and the geography of the International Cartographic Association". In: International Journal of Cartography 3.sup1, pp. 9–31. DOI: 10.1080/23729333.2017.1288535.
- Kraak, Menno-Jan, Britta Ricker, and Yuri Engelhardt (Dec. 2018). "Challenges of mapping sustainable development goals indicators data". In: *ISPRS International Journal of Geo-Information* 7.12, p. 482. DOI: 10.3390/ijgi7120482.
- Kraak, Menno-Jan, Robert E. Roth, et al. (2020). *Mapping for a sustainable world*. New York, New York, USA: The United Nations, p. 79.
- Kray, Christian and Auriol Degbelo (2019). "Map plasticity". In: Poster Session at Conference on Spatial Information Theory: COSIT'19. Regensburg, Germany. URL: http://doi.org/10.5281/zenodo.3385238.
- Kray, Christian, Falko Schmid, and Holger Fritze (July 2017). "Guest editorial: map interaction". In: *GeoInformatica* 21.3, pp. 573–576. DOI: 10.1007/s10707-016-0290-x.
- Krueger, Charles W. (June 1992). "Software reuse". In: ACM Computing Surveys 24.2, pp. 131–183. DOI: 10.1145/130844.130856.
- Kuhn, Werner (2012). "Core concepts of spatial information for transdisciplinary research". In: *International Journal of Geographical Information Science* 26.12, pp. 2267–2276. DOI: 10.1080/13658816.2012.722637.
- Kuhn, Werner, Tomi Kauppinen, and Krzysztof Janowicz (2014). "Linked Data A paradigm shift for Geographic Information Science". In: *Geographic Information Science Eighth International Conference*. Ed. by Matt Duckham et al. Vienna, Austria: Springer International Publishing, pp. 173–186. DOI: 10.1007/978-3-319-11593-1_12.
- Lacoche, Jérémy et al. (2019). "3DPlasticToolkit: plasticity for 3D user interfaces".
 In: Virtual Reality and Augmented Reality 16th EuroVR International Conference (EuroVR 2019). Ed. by Patrick Bourdot et al. Tallinn, Estonia: Springer, Cham, pp. 62–83. DOI: 10.1007/978-3-030-31908-3_5.
- Lai, Pei-Chun and Auriol Degbelo (June 2021). "A comparative study of typing and speech for map metadata creation". In: *Proceedings of the 24th AGILE Conference* on Geographic Information Science (AGILE 2021). Ed. by Panagiotis Partsinevelos,

Phaedon Kyriakidis, and Marinos Kavouras, pp. 1–12. DOI: 10.5194/agile-giss-2-7-2021.

- Lakoff, George and Mark Johnson (Jan. 2003). *Metaphors we live by.* London, United Kingdom: The university of Chicago Press.
- Larkin, Jill H. and Herbert A. Simon (Jan. 1987). "Why a diagram is (sometimes) worth ten thousand words". In: *Cognitive Science* 11.1, pp. 65–100. DOI: 10.1111/j.1551-6708.1987.tb00863.x.
- Lemmens, Rob et al. (Sept. 2006). "Integrating semantic and syntactic descriptions to chain geographic services". In: *IEEE Internet Computing* 10.5, pp. 42–52. DOI: 10.1109/MIC.2006.106.
- Library of Congress (2020). MARC Standards (Network Development and MARC Standards Office, Library of Congress). URL: https://www.loc.gov/marc/ (visited on 09/02/2020).
- Lutz, M. and E. Klien (Mar. 2006). "Ontology-based retrieval of geographic information". In: International Journal of Geographical Information Science 20.3, pp. 233– 260. DOI: 10.1080/13658810500287107.
- MacEachren, Alan (1995). How maps work: representation, visualization, and design. Guilford Press.
- MacEachren, Alan and Menno-Jan Kraak (2001). "Research challenges in geovisualization". In: Cartography and Geographic Information Science 28.1, pp. 3–12. DOI: 10.1559/152304001782173970.
- Mai, Gengchen et al. (July 2020). "Semantically-enriched search engine for Geoportals: A case study with ArcGIS online". In: Proceedings of the 23th AGILE Conference on Geographic Information Science (AGILE 2020). Ed. by Panagiotis Partsinevelos, Phaedon Kyriakidis, and Marinos Kavouras. Vol. 1, pp. 1–17. DOI: 10.5194/agilegiss-1-13-2020. URL: https://agile-giss.copernicus.org/articles/1/13/ 2020/.
- Marzouki, A et al. (2017). "The relevance of geovisualization in citizen participation processes". In: Proceedings of the 18th Annual International Conference on Digital Government Research. Ed. by Charles C. Hinnant and Adegboyega Ojo. dg.o '17. Staten Island, NY, USA: ACM, pp. 397–406. DOI: 10.1145/3085228.3085240.
- Meixner, Gerrit, Gaëlle Calvary, and Joëlle Coutaz (2014). Introduction to modelbased user interfaces. URL: http://www.w3.org/TR/2014/NOTE-mbui-intro-20140107/.
- Mikkonen, Tommi and Antero Taivalsaari (May 2019). "Software reuse in the era of opportunistic design". In: *IEEE Software* 36.3, pp. 105–111. DOI: 10.1109/MS. 2018.2884883.
- Monmonier, Mark (2005). "Lying with maps". In: *Statistical Science* 20.3, pp. 215–222. DOI: 10.1214/08834230500000241.
- Mons, Barend et al. (Mar. 2017). "Cloudy, increasingly FAIR; revisiting the FAIR Data guiding principles for the European Open Science Cloud". In: Information Services & Use 37.1, pp. 49–56. DOI: 10.3233/ISU-170824.
- Muehlenhaus, Ian (Nov. 2013). "The design and composition of persuasive maps". In: *Cartography and Geographic Information Science* 40.5, pp. 401–414. DOI: 10.1080/15230406.2013.783450.
- Mwalongo, F. et al. (June 2016). "State-of-the-art report in web-based visualization". In: Computer Graphics Forum 35.3, pp. 553–575. DOI: 10.1111/cgf.12929.
- Mylavarapu, Pranathi et al. (2019). "Ranked-list visualization: A graphical perception study". In: Proceedings of the 2019 CHI Conference on Human Factors in Comput-

ing Systems - CHI '19. Ed. by Stephen A. Brewster et al. Glasgow, Scotland, United Kingdom: ACM Press, p. 192. DOI: 10.1145/3290605.3300422.

- Noy, Natasha, Matthew Burgess, and Dan Brickley (2019). "Google Dataset Search: Building a search engine for datasets in an open Web ecosystem". In: *Proceedings* of The Web Conference (WebConf'2019). Ed. by Ling Liu et al. San Francisco, California, USA: ACM Press, pp. 1365–1375. DOI: 10.1145/3308558.3313685.
- Obrenovic, Zeljko, Julio Abascal, and Dusan Starcevic (May 2007). "Universal accessibility as a multimodal design issue". In: *Communications of the ACM* 50.5, pp. 83–88. DOI: 10.1145/1230819.1241668.
- Pelzetter, Jens (Apr. 2020). "A declarative model for accessibility requirements". In: Proceedings of the 17th International Web for All Conference. Ed. by Carlos Duarte et al. Taipei, Taiwan: ACM, pp. 1–10. DOI: 10.1145/3371300.3383339.
- Peuquet, Donna J. (Sept. 1988). "Representations of geographic space: toward a conceptual synthesis". In: Annals of the Association of American Geographers 78.3, pp. 375–394. DOI: 10.1111/j.1467-8306.1988.tb00214.x.
- Plaisant, C., J.-D. Fekete, and G. Grinstein (Jan. 2008). "Promoting insight-based evaluation of visualizations: from contest to benchmark repository". In: *IEEE Trans*actions on Visualization and Computer Graphics 14.1, pp. 120–134. DOI: 10.1109/ TVCG.2007.70412.
- Purves, Ross S., Paul Clough, Christopher B Jones, Avi Arampatzis, et al. (Aug. 2007). "The design and implementation of SPIRIT: a spatially aware search engine for information retrieval on the Internet". In: International Journal of Geographical Information Science 21.7, pp. 717–745. DOI: 10.1080/13658810601169840.
- Purves, Ross S., Paul Clough, Christopher B Jones, Mark H. Hall, et al. (2018). "Geographic information retrieval: progress and challenges in spatial search of text". In: *Foundations and Trends® in Information Retrieval* 12.2-3, pp. 164–318. DOI: 10.1561/150000034.
- Ren, Donghao, Tobias Hollerer, and Xiaoru Yuan (Dec. 2014). "iVisDesigner: expressive interactive design of information visualizations". In: *IEEE Transactions on Visualization and Computer Graphics* 20.12, pp. 2092–2101. DOI: 10.1109/TVCG. 2014.2346291.
- Roberts, Jonathan C. (2008). "Coordinated multiple views for exploratory geovisualization". In: *Geographic Visualization*. Ed. by Martin Dodge, Mary McDerby, and Martin Turner. John Wiley & Sons, Ltd, pp. 25–48. DOI: 10.1002/9780470987643. ch3.
- Robinson, Anthony C. (Jan. 2019). "Representing the presence of absence in Cartography". In: Annals of the American Association of Geographers 109.1, pp. 286–300. DOI: 10.1080/24694452.2018.1473754.
- Robinson, Anthony C., Urška Demšar, et al. (Oct. 2017). "Geospatial big data and cartography: research challenges and opportunities for making maps that matter". In: *International Journal of Cartography* 3.sup1, pp. 32–60. DOI: 10.1080/23729333. 2016.1278151.
- Robinson, Anthony C., Scott Pezanowski, et al. (Nov. 2013). "Symbol Store: sharing map symbols for emergency management". In: *Cartography and Geographic Information Science* 40.5, pp. 415–426. DOI: 10.1080/15230406.2013.803833.
- Roth, Robert E. (2013a). "An empirically-derived taxonomy of interaction primitives for interactive cartography and geovisualization". In: *IEEE Transactions on Visualization and Computer Graphics* 19.12, pp. 2356–2365. DOI: 10.1109/TVCG.2013. 130.

- Roth, Robert E. (2013b). "Interactive maps: What we know and what we need to know". In: *Journal of Spatial Information Science* 6, pp. 59–115. DOI: 10.5311/JOSIS.2013.6.105.
- (Mar. 2017). "Visual variables". In: International Encyclopedia of Geography: People, the Earth, Environment and Technology. Ed. by Douglas Richardson et al. Oxford, UK: John Wiley & Sons, Ltd, pp. 1–11. DOI: 10.1002/9781118786352. wbieg0761.
- Roth, Robert E., Cynthia A. Brewer, and Michael S. Stryker (Mar. 2011). "A typology of operators for maintaining legible map designs at multiple scales". In: *Cartographic Perspectives* 68, pp. 29–64. DOI: 10.14714/CP68.7.
- Roth, Robert E., Arzu Çöltekin, et al. (Oct. 2017). "User studies in cartography: opportunities for empirical research on interactive maps and visualizations". In: *International Journal of Cartography* 3.sup1, pp. 61–89. DOI: 10.1080/23729333. 2017.1288534.
- Safarov, Igbal, Albert Meijer, and Stephan Grimmelikhuijsen (May 2017). "Utilization of open government data: A systematic literature review of types, conditions, effects and users". In: *Information Polity* 22.1, pp. 1–24. DOI: 10.3233/IP-160012.
- Saket, Bahador, Alex Endert, and Cagatay Demiralp (July 2019). "Task-based effectiveness of basic visualizations". In: *IEEE Transactions on Visualization and Computer Graphics* 25.7, pp. 2505–2512. DOI: 10.1109/TVCG.2018.2829750.
- Satyanarayan, Arvind and Jeffrey Heer (June 2014). "Lyra: An interactive visualization design environment". In: Computer Graphics Forum 33.3, pp. 351–360. DOI: 10.1111/cgf.12391.
- Satyanarayan, Arvind, Bongshin Lee, et al. (2020). "Critical reflections on visualization authoring systems". In: *IEEE Transactions on Visualization and Computer Graphics*, pp. 1–1. DOI: 10.1109/TVCG.2019.2934281.
- Satyanarayan, Arvind, Dominik Moritz, et al. (Jan. 2017). "Vega-Lite: A grammar of interactive graphics". In: *IEEE Transactions on Visualization and Computer Graphics* 23.1, pp. 341–350. DOI: 10.1109/TVCG.2016.2599030.
- Scheider, Simon, Auriol Degbelo, et al. (2014). "Content and context description -How linked spatio-temporal data enables novel information services for libraries". In: gis.Science 04, pp. 138–149.
- Scheider, Simon, Jim Jones, et al. (2014). "Encoding and querying historic map content". In: The 17th AGILE International Conference on Geographic Information Science - Connecting a Digital Europe Through Location and Place. Ed. by J Huerta, S Schade, and C Granell. Castellón, Spain: Springer International Publishing, pp. 251–273. DOI: 10.1007/978-3-319-03611-3_15.
- Schmidt, Douglas C (1999). "Why software reuse has failed and how to make it work for you". In: C++ Report 11.1.
- Schnabel, Olaf (2005). "Map Symbol Brewer a new approach for a cartographic map symbol generator". In: Proceedings of the 22nd International Cartographic Conference. A Coruña, Spain.
- Schnur, Susan, Kenan Bektaş, and Arzu Çöltekin (May 2018). "Measured and perceived visual complexity: a comparative study among three online map providers".
 In: *Cartography and Geographic Information Science* 45.3, pp. 238–254. DOI: 10. 1080/15230406.2017.1323676.
- Schnürer, Raimund et al. (July 2020). "An empirical evaluation of three-dimensional pie charts with individually extruded sectors in a geovisualization context". In: *Information Visualization* 19.3, pp. 183–206. DOI: 10.1177/1473871619896103.

- Seo, Daeil and Byounghyun Yoo (July 2020). "Interoperable information model for geovisualization and interaction in XR environments". In: International Journal of Geographical Information Science 34.7, pp. 1323–1352. DOI: 10.1080/13658816. 2019.1706739.
- Shneiderman, Ben (May 2000). "Universal usability". In: Communications of the ACM 43.5, pp. 84–91. DOI: 10.1145/332833.332843.
- Stolte, C., D. Tang, and P. Hanrahan (2002). "Polaris: a system for query, analysis, and visualization of multidimensional relational databases". In: *IEEE Transactions on* Visualization and Computer Graphics 8.1, pp. 52–65. DOI: 10.1109/2945.981851.
- Swienty, Olivier et al. (Aug. 2008). "The role of relevance and cognition in attentionguiding geovisualisation". In: *The Cartographic Journal* 45.3, pp. 227–238. DOI: 10.1179/000870408X311422.
- Tait, Alex (Apr. 2018). "Visual hierarchy and layout". In: Geographic Information Science & Technology Body of Knowledge. Ed. by John P. Wilson. Q2. DOI: 10. 22224/gistbok/2018.2.4.
- Thevenin, David and Joëlle Coutaz (1999). "Plasticity of user interfaces: framework and research agenda". In: *Proceedings of INTERACT '99.* Ed. by M Angela Sasse and Chris W Johnson. Edinburgh, Scotland, UK: IOS Press, pp. 110–117.
- Thrash, Tyler et al. (2019). "The future of geographic information displays from GI-Science, cartographic, and cognitive science perspectives (Vision Paper)". In: 14th International Conference on Spatial Information Theory (COSIT 2019). Ed. by Sabine Timpf et al. Regensburg, Germany: Schloss Dagstuhl-Leibniz-Zentrum fuer Informatik, 19:1–19:11. DOI: 10.4230/LIPIcs.COSIT.2019.19.
- Trilles, Sergio et al. (Jan. 2020). "Interactive guidelines: public communication of databased research in cities". In: *PLOS ONE* 15.1. Ed. by Sara Rubinelli, e0228008. DOI: 10.1371/journal.pone.0228008.
- Van Ho, Q. et al. (Jan. 2012). "A web-enabled visualization toolkit for geovisual analytics". In: *Information Visualization* 11.1, pp. 22–42. DOI: 10.1177/ 1473871611425870.
- Vanhulst, Pierre et al. (2018). "Designing a classification for user-authored annotations in data visualization". In: Proceedings of the 13th International Joint Conference on Computer Vision, Imaging and Computer Graphics Theory and Applications (VISI-GRAPP 2018) - Volume 3: IVAPP. Ed. by Alexandru Telea, Andreas Kerren, and José Braz. Funchal, Portugal: SCITEPRESS - Science and Technology Publications, pp. 85–96. ISBN: 978-989-758-289-9. DOI: 10.5220/0006613700850096.
- Varanka, Dalia and Lynn Usery (May 2018). "The map as knowledge base". In: International Journal of Cartography 4.2, pp. 201–223. DOI: 10.1080/23729333.2017. 1421004.
- Viegas, Fernanda B. et al. (Nov. 2007). "ManyEyes: a site for visualization at Internet scale". In: *IEEE Transactions on Visualization and Computer Graphics* 13.6, pp. 1121–1128. DOI: 10.1109/TVCG.2007.70577.
- Vilches-Blázquez, Luis M. and Jhonny Saavedra (Feb. 2019). "A framework for connecting two interoperability universes: OGC Web Feature Services and Linked Data". In: *Transactions in GIS* 23.1, pp. 22–47. DOI: 10.1111/tgis.12496.
- Volz, Julius et al. (2009). "Discovering and maintaining links on the web of data". In: 8th International Semantic Web Conference (ISWC 2009). Ed. by Abraham Bernstein et al. Chantilly, Virginia, USA: Springer, Berlin, pp. 650–665. DOI: 10. 1007/978-3-642-04930-9_41.

- Wagemans, Johan et al. (2012). "A century of Gestalt psychology in visual perception II. Conceptual and theoretical foundations." In: *Psychological Bulletin* 138.6, pp. 1218–1252. DOI: 10.1037/a0029334.
- Weikum, Gerhard, Johannes Hoffart, and Fabian Suchanek (2019). "Knowledge harvesting: achievements and challenges". In: *Computing and Software Science: State of the Art and Perspectives*. Ed. by Bernhard Steffen and Gerhard Woeginger, pp. 217– 235. DOI: 10.1007/978-3-319-91908-9_13.
- Wickham, Hadley et al. (Nov. 2010). "Graphical inference for infovis". In: *IEEE Transactions on Visualization and Computer Graphics* 16.6, pp. 973–979. DOI: 10.1109/TVCG.2010.161.
- Wilkinson, Mark D., Michel Dumontier, et al. (2016). "The FAIR guiding principles for scientific data management and stewardship". In: *Scientific Data* 3.160018. DOI: 10.1038/sdata.2016.18.
- Wilkinson, Mark D., Susanna-Assunta Sansone, et al. (Dec. 2018). "A design framework and exemplar metrics for FAIRness". In: *Scientific Data* 5.1, p. 180118. DOI: 10.1038/sdata.2018.118.
- Wills, Graham and Leland Wilkinson (Jan. 2010). "AutoVis: automatic visualization". In: Information Visualization 9.1, pp. 47–69. DOI: 10.1057/ivs.2008.27.
- Wongsuphasawat, Kanit et al. (Jan. 2016). "Voyager: exploratory analysis via faceted browsing of visualization recommendations". In: *IEEE Transactions on Visualiza*tion and Computer Graphics 22.1, pp. 649–658. DOI: 10.1109/TVCG.2015.2467191.
- Yeates, Lindsay Bertram (2004). "Thought experimentation: A cognitive Approach". Doctoral dissertation. University of New South Wales.
- Zhang, Yun et al. (2020). "A decentralized model for spatial data digital rights management". In: *ISPRS International Journal of Geo-Information* 9.2. DOI: 10.3390/ ijgi9020084.
- Zhao, Tian, Chuanrong Zhang, and Weidong Li (2016). "Accessing distributed WFS data through a RDF query interface". In: International Conference on GIScience (GISCience 2016) - Short Paper Proceedings. Vol. 1. 1. Montreal, Quebec, Canada.